



Sensor Technologies to Manage the Physiological Traits of Chronic Pain: A Review

David Naranjo-Hernández *D, Javier Reina-TosinaD, Laura M. RoaD

Biomedical Engineering Group, University of Seville, 41092 Seville, Spain; jreina@us.es (J.R.-T.); lroa@us.es (L.M.R.)

* Correspondence: dnaranjo@us.es; Tel.: +34-954-485-975

Received: 31 October 2019; Accepted: 5 January 2020; Published: 8 January 2020



MDP

Abstract: Non-oncologic chronic pain is a common high-morbidity impairment worldwide and acknowledged as a condition with significant incidence on quality of life. Pain intensity is largely perceived as a subjective experience, what makes challenging its objective measurement. However, the physiological traces of pain make possible its correlation with vital signs, such as heart rate variability, skin conductance, electromyogram, etc., or health performance metrics derived from daily activity monitoring or facial expressions, which can be acquired with diverse sensor technologies and multisensory approaches. As the assessment and management of pain are essential issues for a wide range of clinical disorders and treatments, this paper reviews different sensor-based approaches applied to the objective evaluation of non-oncological chronic pain. The space of available technologies and resources aimed at pain assessment represent a diversified set of alternatives that can be exploited to address the multidimensional nature of pain.

Keywords: objective pain assessment; chronic pain; body sensor networks; heart rate variability; accelerometry; skin conductance, electromyogram; image processing

1. Introduction

Pain is an undesired feeling that can produce a wide range of responses: from mild, localized ache to distress [1-3]. Despite its displeasure, it is one of the vital body alarm systems, as it allows the subject to recognize stimuli of potentially harmful intensities [4]. Pain may be located on a specific area, such as an injury, but it can also be more diffuse, as it occurs with many diseases [3]. Pain is driven by specific nerve fibers that communicate the impulses to the brain, where conscious appreciation takes place, influenced by multiple external factors [3,5].

In contrast to short-lasting acute pain, chronic pain persists even after the cessation of the initial cause and often loses its protective value [1,4,6]. By definition, chronic pain lasts over 3-6 months [7] and is not solved with treatment [8]. Its origin may be nociceptive or neuropathic [7,9], and it can be triggered by various causes (chronic musculoskeletal pain [10,11], chronic low back pain [12], fibromyalgia [13], tension-type headache [14], osteoarthritis [15], whiplash [16], heart and respiratory systems [7,17], endometriosis [18], etc.). Once developed, it may become resistant to standard treatments, greatly affecting patient quality of life [4].

Chronic pain is an important health challenge, as it represents 80% of physician visits [8]. 10% to 55% people in the world suffer chronic pain [7] (19% adults in Europe and 30% in the United States) [11]. It has been associated with great morbidity [8] and has also been proved to cause depression, neuroticism, sleep disorders, and anxiety [7]. Among the elderly population, it is a common impairment with a significant impact on physical and psychological state, quality of life, and even a barrier to social inclusiveness [19].

Pain assessment and management are essential issues to provide interventions to a wide range of disorders [1,20]. However, as pain is a subjective experience, its measurement is difficult [21]. Sensitivity to pain is manifold in the different individuals [22], in part due to its association with motivational and emotional factors (depression, anger, anxiety, resentment), behavioural components, the influence of life-style (marital or workplace distress, etc.), personal skills, capacity to face pain, health beliefs, etc. [3]. Pain is a very personal sensation that is difficult to analyze, understand and explain, moreover if there is a lack of proper communication with the patient [23].

The commonly accepted method to measure pain in adults is based on self-reports and subjective reports from patients, caregivers or medical staff [20,21]. These reports are frequently complemented with physiological measurements, behavioural observations, interviews, medical records, and psychometric measures, among others [3].

The visual analog scale (VAS) and McGill pain questionnaire (MPQ) are the most popular self-evaluation instruments to measure pain intensity in clinical and research environments [3,24,25]. VAS is a graphical method that requests the respondent to self-evaluate her/his pain in a 10cm-length horizontal line with diverse pain legends, with scores ranging from 0 (there is no pain) to 10 (the worst possible pain) (see Figure 1). MPQ defines a questionnaire with words grouped into 20 subclasses [26]. The interviewer asks the respondent to choose the word in each class that better matches her/his pain. If no word defines the pain in a class, he/she will not make a choice. Each word has a value, and the overall summation provides the pain rating index (PRI). Present pain intensity (PPI) is obtained by combining the numbers of the chosen words.



Figure 1. visual analog scale (VAS) pain scale.

While these tools are convenient and useful, subjective reports present several limitations [20], as they include inconsistent metrics, are affected by reactivity to suggestion, efforts in the control of impression or deception, and are prone to differences in the conceptualization of pain represented by the medical staff and patients. Moreover, these methods cannot be applied to children or patients with neurologic disorders, dementia or transient states of consciousness [20].

An inaccurate and inconsistent assessment of pain can lead to an inappropriate and insufficient therapy [27]. Consequently, instead of relief, the patient can experience a physical worsening, physiopathologic effects (increase of blood pressure, heart rate, etc. [19]), and exacerbate her/his psychological distress [27].

A better assessment of pain would allow the personalization of diagnostic and treatment interventions in chronic pain disorders [28], and it is essential to identify novel therapeutic approaches [4]. The objective inspection of pain has received a great attention during the last years, mainly due to the benefits that it can provide to healthcare (e.g., the election of the most appropriate therapy to improve the quality of life [27], patient follow-up in intensive care units [29], reduce an over-stimulation in rehabilitation, optimize the administration of anesthesia [30] and the dosage of analgesics [23]), and in particular to cases where patients cannot describe their pain in the clinical context and daily life (drowsiness, dementia, etc.) [23]. For these reasons, the accurate measurement of pain is still a challenge for researchers, healthcare professionals, the legal community and patients with chronic pain [28].

The objective assessment of chronic pain is a difficult problem that can be of great interest in multiple clinical areas to procure a better patient intervention and an improvement in the quality of

life of chronic patients of different etiology. It has been conducted in the bibliography from different points of view, ranging from video signal processing [29,31,32] or the analysis of bio-physiological signals [33–36], to systems combining heterogeneous information sources [3,11,27,37–40]. However, to date a universal method has been developed neither for the objective evaluation of pain [22], nor for the measurement of its seriousness, due to the subjective nature of pain perception by the sufferer, although it is a critical demand for multiple disorders. In the absence of objective measurement reference standards, the evaluation is carried out through subjective self-reports from patients, caregivers or medical staff. It should be noted that despite the lack of an objective reference for the evaluation of chronic pain, many studies have tried to induce pain levels in a controlled manner (by heat [23,41–53], cold [45,48,52], pressure [44] or electrical stimulation [54]) as a first approximation to a method that allows the evaluation of chronic pain [1]. In contrast to the studies based on research with phasic pain stimuli (short duration), the number of studies based on tonic pain stimuli is growing as an approach to chronic pain [23,41–52,52–55]. This validation approach represents an objective way for assessing chronic pain, leaving aside all the subjectivity that may be related to the perception of pain by the chronic patient [23,41–52,52–55]. In addition, it becomes the most appropriate option in the case of patients with absence of communication skills, whether verbal, written or tactile (infants, unconscious patients, critically sick, patients with intellectual impairment, advanced dementia, etc. [45]). Therefore, and given the importance and benefits of the objective evaluation of chronic pain for the scientific community in general, and patients in particular, a great research effort must be carried out from multiple technical and clinical points of view, given the interdisciplinary nature of pain assessment. In this sense, the present work provides a comprehensive overview of sensor technologies and physiological aspects related to the underlying technologies for chronic pain assessment, whether of musculoskeletal, gastrointestinal, fibromyalgia, arthritis, migraine, or other types of chronic pain of non-oncological origin.

Regarding the technological alternatives for pain assessment, it is worth mentioning that portable body sensor technologies have experienced a rapid growth in recent years [56–61]. These technologies allow remote monitoring of biomedical parameters (body temperature, heart and respiratory rate, physical activity, etc.) in a non-obstructive way thanks to the wireless communications of the body sensor networks (BSN) [62–67]. As the next sections report, some of these signs can be used for the objective evaluation of pain. On the other hand, numerous investigations have been carried out to incorporate BSNs in healthcare not only for the remote monitoring of patients, but also due to the advantages offered by biomedical data processing methods and techniques and information fusion from different sensors applied to the early detection of diseases and health adverse events [68–72], including pain management. In addition, increasing the awareness of citizens about their own health state as a result of the use of sensors allows them to take a more proactive attitude, favoring self-care and promoting healthier habits and lifestyles. BSNs are thus contributing towards a paradigm shift from a curative and reactive medicine, to a more preventive, proactive and personalized medicine [62–64,73–75].

This paper reviews the different methods described for the objective evaluation of pain. In particular, it is focused on non-oncologic chronic pain. Section 2 describes the relation and use of heart rate variability for the analysis of chronic pain. Section 3 reveals the interest of accelerometry and its application to the evaluation of physical activity in the study of chronic pain. The utility of skin conductance for the evaluation of pain is analyzed in Section 4. Section 5 reports different examples of the use of biopotentials to analyze chronic pain. Different computer vision techniques have also been employed for pain recognition and Section 6 reviews some significant examples from the literature. Section 7 compiles other sensor technologies that have been applied to the assessment of chronic pain and analyzes the benefits of data fusion. Section 9 reports some examples of digital resources (databases and mobile apps) that have been developed in the context of chronic pain detection. Finally, Section 10 draws the conclusions and future research lines.

2. Heart Rate Variability and Chronic Pain

In the ancient Greece, Aristotle suggested that pain was perceived by the soul, which was located in the heart [76]. Today, it is indisputable that the nociceptive information is processed by the brain, however, extensive interactions between the neural structures involved in pain feeling and heart autonomous control can be observed [77]. As heart rate is affected by emotional activity, this indicator can be used to distinguish between positive and negative emotions [78,79].

The autonomic nervous system (ANS) plays a key role in the regulation of heart rate [79,80]. The parasympathetic nervous system makes heart rate and in turn blood pressure to decrease, whereas the sympathetic nervous system favors the antagonistic action, increasing heart rate and blood pressure.

The sympathetic system prepares us to face situations requiring energy expenditure: an emergency threatening the body internal equilibrium, a sudden and intense exercise, an aggression or temperature change, stress situation, etc. The parasympathetic system stimulates activities that take place in normal conditions to ensure long-term wellness (e.g., digestion) and ease energy storage or saving. Both systems are antagonistic and may not be activated simultaneously, such that the action of any of them "restrains" the other. The interaction between these systems is known as the sympathetic-vagal balance of the ANS [80].

A sign of ANS balance and its conditions is heart rate variability (HRV) [81–84], which refers to short- and long-term variations of heart rate due to several causes. HRV records the variation of beat-to-beat time intervals, also known as R–R intervals, as heart beats are usually associated with the R wave of the electrocardiagram (ECG) [82,85]. An average heart rate of 60 beats per minute does not mean that the interval between successive heart beats is exactly 1.0 s, since it can experience variations in the range between 0.5 s and 2.0 s, with HRV indicating the fluctuations of heart rate around its average value (see Figure 2).



Figure 2. Variations of R-R intervals in an electrocardiogram (ECG) signal.

The ECG signal is an expression of the contractile activity of heart, which can be registered on the breast surface or limbs. Although recording ECG in the limbs presents less drawbacks, it is more sensitive to artifacts [43]. From ECG signals, HRV is measured using long- and short-term recordings. The former use to last 24 hours and are acquired with a Holter, while the latter take 2 to 5 min and are based on a dynamic ECG [86]. Although HRV is commonly established through ECG, it can also be obtained with photopletismography [87].

Time-domain and frequency-domain methods can be distinguished to measure HRV. The former are based on the statistical analysis of the time distance between R waves of the ECG signal (R–R intervals, see Figures 2 and 3). However, their capabilities are limited in terms of specificity and sensibility, and they require long recording periods [86].



Figure 3. Example of a signal obtained from ECG R-R intervals.

As aforementioned, heart rate is not fixed and some ANS-related processes induce variations that are oscillatory or repeated periodically. Frequency-domain methods or HRV spectral estimation identify the spectral components (oscillatory processes) implicit in HRV [82]. To that aim, an estimation of power spectral density (PSD) of the HRV signal is usually performed. As Figure 4 shows, the range between 0.0033 and 0.04 Hz identifies very-low-frequency (VLF) spectral components [88]; the 0.04–0.15 Hz range identifies low-frequency (LF) components; and 0.15-0.4 Hz are associated with the high-frequency (HF) range [89]. These components provide clinical information about the variation of heart sinus rhythm. The LF components are influenced both by the sympathetic and parasympathetic system, although they can also be associated with baroreflex sensitivity [90]. The HF components are related to the parasympathetic system, hence the LF/HF ratio provides information about the sympathetic-vagal balance [90].



Figure 4. Spectrum of a HRV signal.

HRV may be affected by situations requiring a mental effort or stress [19]. A low HRV can be associated with a relaxing state, while the increase of HRV is an indication of potential stress, pain or mental frustration [43]. The vagus nerve influences pain modulation [91] in such a way that chronic pain is associated with an alteration of the descendent inhibitory pathway. As HRV is an approximate sign of of vagal activity, this variable can be related to a dysfunction of the descendent inhibitory pathway. Results reported in [91] support this effect. There are many studies relating chronic pain of different etiology (headache, neck, shoulder, lumbar, hip, etc.) with a dysfunction of the ANS, and specifically with a low HRV mediated by a reduction of the parasympathetic cardiac activity [77,92–97] and a sympathetic activation [91]. This relation has also been evidenced in trials with animals [98].

As the frequency components of HRV allow a detailed analysis of the sympathetic–vagal balance, its study is useful for the evaluation of chronic pain. In [99], a method was developed to assess the intensity of pain through a spectral analysis of HRV fluctuations [100]. An analysis of 51 trials evaluating the influence of chronic pain in the HRV spectrum [95] showed a consistent (moderate to high) effect of reduction of the HF component of HRV in patients with chronic pain, what implies a reduction of parasympathetic activity. This effect is even more obvious in trials addressing fibromyalgia. As aforementioned, persons with chronic pain have a reduced HRV compared with no pain, with differences specially relevant in the HF component [77,95].

Next, results reported in different trials about the assessment of pain through HRV are grouped depending on the etiology of condition, and its application to diverse chronic pain treatments.

2.1. Pain of Musculoskeletal Origin and HRV

The analysis reported in [22] using HRV reveals a reduction of the parasympathetic activity and an increase of the sympathetic activity in patients with chronic pain of musculoskeletal origin.

According to [101], a reduction of cervical chronic pain is associated with a reduction of the LF component of HRV, an increase of the HF component and a reduction of the LF/HF ratio.

In [102], a trial with patients suffering from shoulder and neck pain reported a significant reduction of the LF/HF ratio after the execution of isometric exercise in the control group, while no significant changes were observed in patients with no pain.

In [103], a trial with patients suffering form sciatic pain referred to backbone surgery showed a higher LF/HF ratio in the control group. Moreover, in these patients a higher HF component was correlated with a higher duration of pain episodes.

2.2. Fibromyalgia and HRV

Fibromyalgia is a chronic pain syndrome characterized by a generalized musculoskeletal pain, painful sensing to pressure in specific points, fatigue, unrelaxed sleep and depressed mood [94]. HRV has been correlated with pain in patients with fibromyalgia and it is sensitive to changes of condition, with potential of becoming a useful biomarker for the diagnosis of this disease [94].

2.3. Gastrointestinal Pain and HRV

HRV is a commonly used index in gastroenterology, as it reflects the unbalance between the primary excitatory and inhibitory body systems [104]. ANS is the main communications pathway between brain and intestine, hence it is of particular interest for gastrointestinal disorders, as it is the case of irritable intestine syndrome, which is characterized by a functional abdominal pain [105]. Defining the gravity of the pain according to its duration and intensity, a lower HRV has been related with a higher gravity of abdominal pain in these patients [28,106,107].

Irritable intestine syndrome was also analyzed in [105]. Results showed a significantly lower HRV in women with persistent pain in comparison with women with controlled pain or no pain, and also with respect to men, independently of their pain state. A spectral analysis of HRV showed a reduction of the HF components, i.e., a reduced parasympathetic "restrain" of the sympathetic activity, either in rest and during stress episodes. This phenomenon can be reflecting a peripheral mechanism, such as an intestinal dysfunction, or a mechanism of the central nervous system, such as pain amplification or deficient emotional self-regulation.

According to [108], a higher duration of the disease (chronicity) is related to a reduction of the LF component of HRV, and a higher seriousness of abdominal pain is related with a decrease of the HF component.

2.4. Burning Mouth Syndrome and HRV

The analysis of HRV reported in [109] highlighted the autonomic instability present in the burning mouth syndrome, a chronic pain characterized by a burning sensation in the tongue or mouth. The application of (near infrared) radiation in the stellate ganglion fixes the autonomic dysfunction, being detectable by HRV, what can be useful to follow the evolution and evaluate the therapeutic effectiveness of this approach.

2.5. Pain Catastrophizing and HRV

Pain catastrophizing is "characterized by the trend to magnify the threatening value of pain stimuli and vulnerable feeling in the context of pain, and a relative incapacity to inhibit thoughts

related to pain before, during or after a painful episode" [110]. According to [111], pain catastrophizing is inversely related to the HF components of HRV, which could be used as an index to assess the capacity of emotion regulation and perception of pain threat. The reduction of the HF components of HRV can be the psychophysiological mechanism underlying a higher pain catastrophizing.

Similar results were reported in [8], where pain anxiety and feelings were rated as significant predictors of HRV in a group of patients with benign pain, including patients with fibromyalgia, exposing its relation with chronic pain and autonomic function.

2.6. Pain Therapies and HRV

Results obtained in [112] show that an anthroposophic therapy can be beneficial to several conditions, such as chronic pain, hypertension and mood disorders, with an impact on HRV (see Figure 5).



Figure 5. Sequence of movements in anthroposophic therapy.

Biofeedback therapy is a trans-disciplinary intervention method for chronic pain [113]. These techniques provide patients with visual or auditive feedback of information in real time related to a biological signal reactive to pain and unknown to the patient [114]. Frequently used biomedical signals are electromyogram recordings from a target muscle, temperature, HRV, respiratory frequency or skin galvanic response. The objective is that the patient can perform an intentional control of the biomedical signal, in order to interrupt the reactivity to pain. Results obtained in [114] show that the use of HRV in biofeedback therapies provide benefits in different measurements of reactivity to pain.

The percutaneous stimulation of auricular vagal nerve has been demonstrated to modulate the body sympathovagal (autonomous) balance, becoming an effective treatment for depression, epilepsy and acute and chronic pain [115]. A trial with primary cervical dystonia patients [116] revealed that the stimulation of the vagal nerve led to a reduction of muscle pain, dystonia symptoms and a subjective recovery of motility, sleep and mood, situations that were correlated with an increase of HRV median from 37.8 to 67.6 ms. The spectral analysis of HRV showed a decrease of HF components, what indicates a higher activity of the parasympathetic activity.

Cervical manipulation, often used for the treatment of musculoskeletal disorders, such as neck pain, has a significant influence on HRV, increasing the parasympathetic response and decreasing the sympathetic tone [117].

In a study conducted with patients suffering from chronic lumbar pain (the most prevalent musculoskeletal disorder [118]) associated with an altered alignment of intervertebral discs [119], the practice of Yoga during three months reduced pain feeling, the LF components of HRV and

respiratory rate, and increased significantly the HF components of HRV. Mindfulness-oriented recovery enhancement (MORE) integrates training in "tasting" natural rewards with mindful training and cognitive re-evaluation. It has a demonstrated effectiveness in the reduction of chronic pain, the misuse of opioids and the desire of being in contact with social support groups [120]. The application of this technique has been associated with an increase of HRV [121].

On the other hand, results reported in [122] reveal that HRV can be an useful index for patients suffering from chronic pain that misuse prescribed opioids to procure a better self-regulation and response to the dosage.

3. Accelerometry-Based Activity Evaluation and Chronic Pain

Another way of identifying the presence of pain is through movements and physical activity, which are essentially different in subjects with chronic pain [25]. It is commonly considered that people who feel impaired and report restrictions in daily-life due to pain are less physically active [123–125]. People with chronic pain of physical origin adopt specific behaviors in response to pain, or when this is expected to occur, to protect the body [11,124]. Anxiety produces a feeling of pain augmentation and/or a higher damage makes people behave in a self-protecting way, e.g., avoiding a particular movement. For example, patients with chronic lumbar pain show a coordination of movements and activity patterns different from healthy subjects [126]. These use to move hip and knees asynchronously when performing a lifting exercise, while patients with lumbar chronic pain use to move hip and knees simultaneously [127]. In [128], the variation of shoulder acceleration during lifting movements was evaluated, and the rate was lower for subjects with pain. Patients who have neck pain have a tendency to move the trunk with less intensity, performing the activities more slowly [129].

Pain responsiveness conducts include protection, hesitation, stiffness and fastening. Four subgroups representing the resistance-avoidance pain regulation response model patterns were analyzed in [130]: resistance to distress, resistance to stress, avoidance of apprehension (to pain) and adaptive responses. Results showed a higher pain, impairment and fatigue in patients belonging to the apprehension avoidance group compared to patients with adaptive responses and, as expected, a lower physical activity. However, despite all these conditions, the measurement of physical activity is a challenge for the treatment, rehabilitation and promotion of healthy habits in patients suffering from non-oncologic chronic pain [124,131,132].

The common methods applied to the evaluation of physical activity in persons with chronic pain are self-evaluation questionnaires, such as the international physical activity questionnaire or Baecke physical activity questionnaire (BPAQ). However, these methods have shown a lack of agreement regarding the estimates of physical activity obtained from sensor devices [133–136]. Moreover, patients with high levels of depression use to underscore their activity level [135].

This scenario justifies current efforts for the development of objective physical activity evaluation systems for the identification of pain-related activities (breast pain, headache, etc.) [137,138], for the evaluation of pain intensity [139–147], for the detection of postures, movements or activities that can produce pain [148–153], or for the improvement of quality of life and a reduction of pain intensity [154–158].

Accelerometry is the common method for the objective evaluation of physical activity in observational studies [159] (see Figure 6). There are several accelerometer-based wearable systems for the continuous monitoring of daily activity, as they allow recording variations in the orientation and an easy detection of movements [160]. Some authors have proposed using the count of accumulated steps as an estimate for physical activity [140]. Other studies refer to the total daily activity minutes [161] or a measurement of the intensity of activity [162]. Most traditional accelerometer devices only record a measurement proportional to the average acceleration in a time lapse [159,163], what may lead to errors in the classification of activity when compared with the reference method [164]. Other proposals have been suggested based on the frequency of postural transitions [165] and estimation of energy

expenditure [166]. Nevertheless, the systems reporting more information about the behavioral patterns and habits of the monitored persons are the daily life activity classifiers [167,168].



Figure 6. Example of uniaxial acceleration acquired with a physical activity monitoring device.

Next, some results related to the evaluation of pain through accelerometry and the study of physical activity are detailed.

3.1. Physical Activity as a Predictor of Pain

The origin of musculoskeletal pain is often associated with inappropriate postures or movements [148,149,169]. Such is the case of the study reported in [150], which suggests a relation between the sitting time during the full day and specifically during work time, and the intensity of neck/ shoulder pain in workers developing manual work. In a similar group of persons, a negative correlation was established between walking time during work and lumbar pain [151].

On the contrary, in health sector workers, the amount of sitting time is inversely related with lumbar pain intensity [152]. In another working area, results reported in [153] showed that grape-pruning activity leads to the adoption of curved and forward folded trunk postures inducing a significant risk of lumbar musculoskeletal disorders.

The work presented in [169] describes a system for the prevention, diagnosis and treatment of neck pain in office workers. The referred system is formed by neck movement sensors and a smartphone that allows system management and feedback with the patient to correct inappropriate postures. The authors of the system proposed in [124] based on portable motion sensors investigated the automatic detection of self-protective behaviors.

3.2. Physical Activity as a Pain Index

Several trials have suggested the use of physical activity as an objective index for pain, based on the fact that the intensity of average movement, the maximum intensity of movement and postures acquired through accelerometry are different in persons with chronic pain compared to healthy persons [139]. Results published in [140] show that it is possible to discriminate patients with complex regional pain syndrome from control subjects using a small set of gait characteristics extracted from accelerometer data during a short set of physical performance tests [170]. In [24], gait using inertial sensors was analyzed in adult subjects with chronic arthralgia after Chikungunya virus infection, finding a correlation of pain with alterations in gait, grip strength and balance.

The intensity of physical activity evaluated through accelerometry has also been applied to the assessment of the intensity of chronic or recurrent sub-acute lumbar pain in adolescents [141], in patients with hip or knee osteoarthritis [142], shoulder pain in persons with type-2 diabetes [143], the detection of walking impairments after abdominal surgery [144], juvenile idiopathic arthritis [145], or chronic pain of musculo-skeletal origin in patients and workers [146,147].

In [171], an abdominal pain detection system is proposed based on the accelerometers integrated in a smartphone. In this system, the patient reproduces in a guided way the palpation movement using the smartphone. A detection algorithm recognizes pain events from the acquired accelerometer information. Pediatric abdominal pain has also been evaluated by means of accelerometry through the assessment of the sleep and physical activity [172]. The relationship between self-scored pain, muscular activity and postural load during cleaning activities was studied in [173], and the authors found a higher muscular activity in the case of low-level pain compared to subjects reporting a higher level of pain.

Accelerometry has also been used to study the relationship of physical activity with pain in patients with knee osteoarthritis, since these patients may become more inactive due to pain and functional limitations [174]. Conversely, the results obtained in [25] indicated an increase in physical activity in patients with osteoarthritis of the facet joint after a paravertebral spinal injection.

In [175], the authors made an objective comparison of physical activity in patients with chronic lumbar pain, who were classified in several groups: avoidance, persistent, mixed executors (significant avoidance and persistent behavior), or functional executors (reduced avoidance and persistent behavior). Physical activity level did not differ among the four groups. However, an association between the pain intensity level and the activity level in the avoidance and persistent groups was found.

In addition to activity intensity, pattern activities experience detectable alterations in persons affected by pain. According to [176], the distribution of activities during the day significantly differ in patients with chronic lumbar pain. This study reported that these patients show higher activity levels during the morning and lower during the night, compared to control subjects.

Although physical activity has been usually associated with lower levels of pain (the lower the activity the lower the inhibitory capacity of pain, and the higher activity the lower pain [177]), some conditions are characterized by the opposite. In persons with knee osteoarthritis, the increase of knee load frequency (steps/day evaluated through accelerometry) and its magnitude (knee adduction angular impulse evaluated through isokinetic dynamometry) has been associated with an increase of pain [178]. The temporal relationship between pain and activity in patients with acute lumbar pain was studied in [179]. According to their results, a high correlation between activity and pain was evidenced during the first week of acute lumbar pain. In most cases, pain persisted 30 min after activity. As patient condition improved and reported less pain, the relation between activity and pain disappeared.

3.3. Benefits of Physical Activity to Pain

Physical activity has also been related to an improved quality-of-life and a reduction of pain intensity in patients with chronic pain. The reduction of pain through physical activity has been shown to be able to improve the ability of people with musculoskeletal pain to perform daily activities [180].

In [154], the authors analyzed physical activity performed by patients with fibromyalgia. Participants in the trial mostly had sedentary habits, characterized by 4019 ± 1530 steps/day. Results showed a linear relation between the increase of day steps and health results.

Another study showing the benefits of performing moderate physical activity for the reduction of neck/shoulder pain in workers and manual labor was reported in [155].

Physical activity has been associated with an improved quality-of-life in persons with multiple sclerosis [156]. This work concluded that the most physically active patients reported lower levels of disability, depression, fatigue and pain.

The study developed in [157] associated a low or moderate physical activity with reductions in the verbal rating scale pain assessment for patients enrolled in a cognitive-behavioral therapy intervention program.

The results reported in [158] suggest that the development of highly intense activities and a high rate of activity fluctuations are associated with a worse sleep in patients with chronic pain. Therefore, the modulation of activity may be a key therapy to address sleep disorders in subjects with chronic pain.

Considering that the benefits of physical activity are dependent on their correct performance, the control of adherence to activity programs at home is of great interest, which can also be evaluated through accelerometry [181]. This is also the purpose of the "selfBACK" m-Health decision support

system, which is introduced in [182,183] for the self-control of lumbar pain. This system controls the activity of the subjects through human activity recognition by using accelerometer sensors to evaluate their adherence to the prescribed physical activity plans.

The authors of another work [184] used an inertial sensor to evaluate the angular velocity and the maximum angle of column movements. Their results confirm the possibility of using this information for a quantitative evaluation of the rehabilitation progress of patients with low back pain.

4. Skin Conductance and Chronic Pain

Skin conductance is a parameter related to the electrical resistance of skin, which increases with sweat [19]. Electrodermal activity is considered a sign of internal tension of a subject, as sweat glands are innervated by the sympathetic branch of the ANS. In presence of a stress situation, pain, emotional excitation or an intense mental effort, skin conductance can experience a sudden increase in a period of 1–3 s. This signal allows the differentiation between conflict and no-conflict situation, or between rage and fear.



Figure 7. Fluctuations of skin conductance registered in [185].

Conductance measurement is based on the injection of an electric current into the body through a pair of electrodes, and the measure of the electric voltage produced as a consequence of the circulation of electric current through the body [186]. According to Ohm's law, the ratio between voltage and current provides an impedance (Z = V/I). This way, impedance (Z) is described as the opposition of a conductor to the electric current flow. This property is dependent on frequency and the particular characteristics of the medium through which current flows, and it is inversely proportional to conductivity (σ).

Results published in [41] show a relationship between the response of skin conductance and pain induced by heat. A study focused on the evaluation of post-surgery pain evidenced a correlation between the number of fluctuations per second of skin conductance and the level of pain (0.07 for no pain, 0.16 for low pain, 0.28 for moderate pain y 0.33 for severe pain) [185,187] (see Figure 7). The problem is that this signal is also influenced by factors such as external temperature, and requires reference measurements and calibration [43].

Skin conductance has also been used for the evaluation of pain in patients with fibromyalgia [188–190], showing a response above the base line [191,192], and also for the evaluation of pain in patients with chronic pain in the back [193–198] or with a high risk of chronic back pain [197], for the evaluation of shoulder pain [199], chronic lumbar pain [200], patients with post-trauma stress disorders [201], higher depressive disorder [202], tensional headache [195,203], migraine [204], irritable intestine syndrome [205], or arthritis [196]. The authors of [206] quantified the response to pain in patients with Rett syndrome by measuring skin conductance.

In [207], the electrodermal reactivity of patients with chronic lumbar pain to emotional speech was analyzed. In these patients, the response of skin conductance induced by pain-related speech was higher compared to neutral speech. This sensitivity to pain-related speech was also evidenced by the study developed in [208]. Similar trials were conducted in [209] and [210], by analyzing reactivity to subjective thoughts of pain perception [209] or video images [210], finding equivalent results.

Patients with chronic pain exhibit impaired cognitive functions, affecting decision making [211]. In the group of patients studied in [212], an association was reported between the lack of an anticipatory response of skin conductance in decision making process, what suggests that these processes are more dependent on cortical resources.

On the other hand, slow and deep respiration techniques, used in diverse relaxation methods for the treatment of chronic pain [213], have reported a significant effect in the reduction of skin conductance [214].

5. Biopotentials for Pain Assessment

The electrical biopotentials generated by the human body are signals commonly used in medical diagnosis. The main biopotentials are electrocardiagram (ECG) [82,85], electromyogram (EMG) [11,35, 43,215] and electroencephalogram (EEG) [46,50,216–219]. Biopotential sensors are usually composed of the following elements: electrodes, used for the transduction of the ionic signals inside the body to electrical signals; an analog conditioning stage for the amplification of the electrical signal, of very low intensity, at measurable levels avoiding electromagnetic interference (EMI); and an analog–digital conversion stage. Isolation barriers to protect the physical integrity of the user are necessary. Research about the biopotential instrumentation is focused on obtaining greater patient comfort, an improvement in portability and ease of use. A problem presented by this type of systems is that the electrodes produce drifts of the continuous level that can be several orders of magnitude above the amplification is based on analog electronic stages applying high-pass signal filtering, distributing the amplification in several stages, one with lower gain before filtering and a final stage of high gain for the adaptation to adequate signal levels. At present, sigma–delta analog–digital conversion technology provides very high accuracy, which reduces the need for gain of the amplification stage.

5.1. Electromyogram Sensors and Chronic Pain

Muscular electric activity is a sign of general psychophysiological activity, in general [35], and in particular, an increase of muscular tone reveals an increase of the sympathetic nervous system, whereas a reduction of somatomotor activity is associated with parasympathetic activation [43]. A high level of muscular tension is a sign of stress [220], which is also expected under a pain situation [35,43].

EMG is a measurement of biopotentials related to the muscle electric activity, which is related to muscular activity or the frequency of muscular tension [220]. For this reason, EMG signals have been used in pain assessment studies [11]. Low back pain is related to alterations of the muscular function of the trunk as a consequence of pathological states of muscular activation by the central nervous system. The measurement of the surface electromyogram (sEMG) has been applied to the study of these alterations [221]. The EMG measurement system proposed in [222] evaluated neck fatigue by providing the user with feedback for posture correction [222]. EMG has also been applied to the evaluation of low back pain [215]. In [223], shoulder and neck pain induced as a physiological response to stress were shown to be associated with a slight increase of heart rate and sEMG activity at the trapecius muscle. In [224], an absence of muscular relaxation response in the flexion movement of the trunk was reported for persons with chronic pain. Moreover, in these subjects a lower muscular activation was evidenced in the contraction phase. As no differences were found in the muscular activation patterns when they did not move, this study suggested the possibility of identifying chronic pain during the execution of physical activities.

Another study recorded the EMG of maxillar muscles in patients with disorders in the temporoflexion exercises, analyzing the flexion/exmandibular joint, showing a higher activity for mandibular activity during rest periods in patients with pain compared to healthy subjects [225]. In [226], EMG was used in tension ratio for the activation of the lumbar para-spinal muscle, the muscular activation for the medium lumbar para-spinal muscle, and the total flexion developed. The use of these indices provided a classification of subjects with pain with a precision of 86%.

In [118], an electrode matrix was used for the acquisition of EMG signals and a Bayesian classifier was proposed as a predictor of lumbar pain (70% success). In this case, Sorenson test was applied to induce muscle fatigue in the lumbar region [118]. The same technique was used in [227], but in this case with an artificial neural network (80% success).

EMG signals were applied in [228] to detect patients with chronic neck pain through the analysis of nociceptive withdrawal reflex in the leg. Nociceptive withdrawal reflex has demonstrated to be a useful tool for the objective evaluation of hyperexcitability of spinal chord present in chronic pain disorders. Under a pain situation, the nociceptive system experiences a change leading to hyperexcitability, i.e., to the presence of pain after innocuous stimulation, or excessive pain after low-intensity nociceptive stimulation. Patients with chronic pain syndrome, such as cervical stabbing syndrome, fibromyalgia, osteoarthritis, tensional cephalea, endometriosis, or chronic lumbar pain show hyperexcitability after stimulation of healthy tissues. Based in this principle, the system proposed in [229] used an electrical stimulation device that generated a train of five pulses of constant current level, with 1-ms length and repeated at a rate of 200 Hz. As a pain threshold, the lowest current amplitude evoking pain sensing was established, setting the stimulation 1.5 times the value of the threshold. The activity of the anterior tibial muscle was recorded using sEMG. Results showed a high success rate for the correct evaluation of chronic patients compared to healthy volunteers (80% success).

EMG signals have also been used in some pain treatment interventions. An example is the study developed in [230], which proposed an EMG system to provide pain relief for patients with patellofemoral pain syndrome. In these patients, pain is caused by an imbalance between the vastus medialis obliquous and vastus lateral. The system compared EMG signals from both muscles and generated a contraction in the vastus lateral through electrical stimulation that compensated the imbalance.

Another work [231] proposed a facial mask with EMG sensors for the evaluation of facial expressions as the basis of an automatic tool for the objective evaluation of human pain.

5.2. Electroencelography and Chronic Pain

EEG is a measurement of biopotentials associated with the electrical activity of the brain by an array of electrodes attached to the scalp. Since EEG signals are very weak, the application of conductive gels can improve the quality of the signal by reducing the impedance of the electrode-skin contact [46]. EEG has numerous practical advantages, since it is a non-invasive technique, with high temporal resolution, provides relevant clinical information, is low-cost and requires little maintenance [50]. EEG studies have shown the activation of specific regions of the brain as a result of pain stimuli [216,217].

Frequency domain analysis, either by fast Fourier transform (FFT), regression models (RM) or wavelet transform (WT) [46], has been employed to study the experience of pain by dynamic analysis of the power in frequency bands [216,218]. Some studies have highlighted the influence of pain in certain frequency bands of the EEG signal, especially in the alpha [50,219] and the gamma [50,232] bands.

Amplitude changes in the alpha frequency band (8–13 Hz) measured at standardized bilateral temporal scalp electrodes (T7 and T8) are associated with tonic pain [216,233]. Power increases in the gamma band (30–100 Hz) have also been related to pain stimuli [216,218]. Since the gamma band is susceptible to higher frequency noise related to EMG signals of facial muscles and artifacts derived from oculomotor activities, adequate signal filtering must be performed [216,234].

In [46], chronic and postoperative pain was analyzed from EEG signals using machine learning techniques. EEG signals were used in [235] (23 channels, a sampling frequency of 256 Hz and 16-bit resolution) to differentiate between harmful (secondary) and benign (primary) headache conditions. EEG has also been employed for the evaluation of chronic pain in general [236] or as an objective marker of pain during the first stage of labor [216].

5.3. Electrocardiagram and Chronic Pain

ECG is a biopotential measurement related to the electrical activity generated by the heart [35]. Usually, ECG signal is measured by two electrodes, one on the upper right and one on the lower left of the body, although a third electrode can be used as reference. ECG signal has a characteristic pattern formed by a P wave, a QRS complex and a T signal. Multiple parameters can be extracted from the characteristics of the previous waves, such as the intervals between QRS complexes (R–R intervals), intervals between P waves (P–P intervals), P–R intervals, ectopic pacemakers, etc. However, given the involvement of HRV in the ANS balance and its direct relationship with pain, these parameters have been largely relegated in favor of HRV in the study of pain. For this reason, the fundamentals and applications of HRV for pain assessment were presented in Section 2.

6. Computer Vision and Chronic Pain

Computer vision or artificial vision is a scientific/technical discipline that deals with methods for acquisition, processing and analysis of physical world images to abstract relevant information to be managed from a computer. These methods include aspects related to geometry, statistics or information theory. Data correspond to image or image sequences acquired from a video camera or other similar system, such as a medical imaging device.

6.1. Recognition of Facial Expressions Associated with Pain

Given that pain response behavior includes facial expressions, possibly intended to request support or empathy [237], computer vision techniques have been used for facial recognition of pain [238]. For this purpose, facial action coding System (FACS) is often used, which is a taxonomy that describes the movements of the facial muscles, to categorize them in a standard and systematic way according to different expressions of emotions, including pain [239]. The implementation of FACS in computer vision systems is performed by detecting geometric features on the face and movement patterns related to facial expressions [240]. Previously, the images must be managed by a procedure for the standardization of image characteristics [54,239]: image segmentation to recognize the face on the background, centering of facial characteristics, normalization of the arguments in the image data, image scaling to predetermined pixel dimensions, etc. If convenient, the image is cropped [241]. The color dimension is usually reduced to black and white. An affine transformation based on face landmarks (mouth corners, nose tip and eyes corners, mainly) is usually employed for face alignment and rotation. The elimination of redundancies and noise from the image allows highlighting the characteristics and structures for a better operation of the facial expression recognition algorithms [239]. Local operations on image texture can also be performed with operators such as local binary patterns (LBP). Discrete cosine transform (DCT) can also be used in the image description process [241]. Other descriptors are based on Taylor series expansion of the image function, named according to the order of the derivative [240,242]. The histogram of the image is a descriptor related to the zero-order coefficient of Taylor series. Other descriptors such as histogram of oriented gradients (HOG) are associated with the first derivative (directional gradient). The speeded up robust features (SURF) and scale-invariant feature transform (SIFT) image descriptors correspond to the second-order derivative of the image (Hessian matrix), which are stable with the image scale and intensity. Finally, the classification is carried out using methods such as those described in Section 8.

The features of face shape have been used for the classification of painful tasks [243] or the evaluation of pain intensity in neonates [244]. The system proposed in [245] used computer vision techniques to analyze the face and the upper part of the body and distinguish among 12 affective states, two of which were fear and anxiety.

A study showed that body and head movements are as effective as facial expressions to detect depression [246]. The system proposed in [247] analyzed images of the movements of sitting and standing, and was able to discriminate between subjects with no pain, subjects with chronic pain,

and subjects simulating pain. Differences among the three classes were found in terms of knee, hip and trunk displacements with respect to mean feet support point. In [248], the movements of vertebral deformation for a group of patients with low-back surgery were evaluated through image analysis.

With this method, a prediction of pain intensity in a 0-10 scale was reported with a precision of 75%.

6.2. Functional Magnetic Resonance Imaging (fMRI) and Chronic Pain

Functional magnetic resonance imaging (fMRI) is another computer vision technique that has been used in the analysis of the central nervous system response in relation to the pain perception and modulation processes, both chronic and acute ones [249,250]. Classification task in these systems is normally based on multivariate pattern analyses (MVPA) [47,251], and on the brain response in two recognizable situations [252]: baseline activity and pain state. Baseline activity is used as reference in the classification method. Magnetic resonance imaging (MRI) sequences are used to construct voxel-based models in activity time series [253]. Images are previously preprocessed to eliminate motion artifacts, breathing and cardiac activity [252]. Through a standardized brain atlas, the brain is segmented into regions of interest (ROIs). The average value of the activity of each ROI is evaluated, and a covariance matrix is constructed [251]. Sparse graphical models (gLASSO) can then be used to reduce the dimensions of the space and estimate the characteristics by partial correlations (inverse of the covariances) [249–251]. Structural properties such as volume, surface, curvature or thickness can then be estimated [254].

The characteristics of white matter (axonal integrity) or gray matter (integrity of neuronal bodies) are related to the structural characteristics of the brain assessed through fMRI [255]. Using this technique as a basis, alterations in white matter have been related to different chronic pain conditions [256,257]. On the other hand, a lower density of gray matter in the dorsolateral prefrontal cortex, a region responsible for the modulation and recognition of pain, has been correlated with the chronic low back pain [257–259]. On the other hand, fibromyalgia has been related to alterations in gray and white matter volume in certain areas of the brain [260]. Another study in older people has shown an association between chronic pain and brain atrophy [44].

6.3. Functional Near-Infrared Spectroscopy (fNIRS) and Chronic Pain

Another neuroimaging method is functional near-infrared spectroscopy (fNIRS) [45]. This method estimates the concentration of oxygenated hemoglobin (HbO) and deoxygenated hemoglobin (HbR) in real time using photoelectric methods. Light absorption rate of HbO and HbR is evaluated at two different wavelengths, typically 695 nm and 830 nm [261]. HbO light absorption is higher than HbR absorption at 830 nm. At 695 nm, just the opposite occurs. From the ratio of light received in the heart pulse related components of both wavelengths it is possible to establish the metabolic changes and the hemodynamics of brain activity [45]. The spatial resolution is obtained through a multichannel optical system that is distributed in matrix form over both hemispheres of the head. Vascular dilation increases brain blood flow, causing the activation of cortical regions. In this way, neuronal activity can be related to fluctuations in the hemodynamic response [262]. Different approaches can be subsequently used for the extraction of neurological characteristics, in time, frequency or time-frequency domains (e.g., wavelets) [48].

fNIRS is a non-invasive technique that has shown to be very useful in the evaluation of brain activity in response to pain stimuli. In [263], fNIRS was proposed as a biomarker of pain in conjunction with machine learning algorithms. Allodynia and motor loss of fibromyalgia patients has been related to alterations in the upper parietal gyrus bilateral region detected by fNIRS [264]. The analysis of cortical activity has shown that the primary motor cortex, premotor cortex and somatosensory cortex are related to pain perception, which was corroborated by pain studies conducted with fMRI, showing the usefulness of HbO as a possible qualitative pain biomarker [265].

7. Other Pain Sensors and Data Fusion

This section compiles other sensing technologies and methods applied to the evaluation of pain, different from those already reviewed, and analyzes the benefits of data fusion.

Respiratory rate sensors have been used for the recognition of emotional states [43]. Negative emotions generally induce an irregular respiratory pattern [78]. Respiratory rate is generally reduced with relaxation, however, exposure to tension situations can lead to the sudden stop of respiration. A fast and deep breathing can denote emotions like rage or fear, but sometimes happiness. A fast surface respiration can indicate a tense situation caused by panic, fear or concentration. Slow and deep respiration indicates a state of relaxation, whereas a slow and surface respiration is associated with abstinence states, depression, and also happiness (under calm situations). This way, respiratory sensors might be used for the evaluation of pain, insofar as breathing influences the emotional state [43,78].

According to [108], the irritable intestine syndrome is associated with a higher blood pressure compared to healthy subjects. A higher duration of the syndrome (chronicity) is related with an increase of systolic arterial pressure. Pain chronicity in these patients has also been associated in [28] with a blood pressure drop, a reduction of tachycardia associated to a lower seriousness of the disease (mechanism of "resistance to pain") and a reduction of bradycardia associated with a higher seriousness of the disease (mechanism of "pain decompensation").

Skin temperature varies during stress episodes, which can occur in presence of pain situations. As muscles are in a low-tension state, blood vessels contract and cause temperature reduction. However, like conductivity, skin temperature is also influenced by external factors, and it responds slowly to emotional state changes [43].

Blood pulse volume is a measure that allows the determination of the quantity of blood circulating in blood vessels. Photopletismography is based on the emission of light on a body area irrigated with blood vessels, usually a finger [43,266]. A photodetector acquires the light reflected, which depends on the blood volume. Therefore, this technique can be used to measure vasoconstriction, which may be related to pain states.

The mechanical impedance (stiffness) of arterial walls is calculated on a beat-to-beat basis through blood pressure and photopletismography measurements. Results reported in [30] suggest that the changes of stiffness levels can be used to quantify pain level. On the other hand, bioimpedance spectrometry (BIS) is an objective non-invasive tool that has been applied to the diagnostic of knee osteoarthritis [267].

In [268], a biosensor was proposed for the evaluation of pain intensity based on the detection of inflammation markers such as Prostaglandine *E*2, Cyclooxygenase (COX - 1 and COX - 2), Leucotrien (LTA_4 and LTB_4) and Troponin-C. These measurements are routinely performed in blood analysis laboratories, using specialized equipment based on enzyme-linked immunosorbent assay (ELISA). This immunoassay technique detects an immobilized antigen through an antibody tied to an enzyme that generates a detectable product or a change of color in the sample. The procedure involves getting a blood sample of the patient, the preparation of reactives, antibody solution, substrate solution, and incubation with light protection, light exposure and correction of wavelength. The process takes approximately one week to obtain the final report. The proposed biosensor requires a blood sample and estimation can be done in a shorter time, although the necessary equipment is complex and costly, as it requires microelectronic technologies, the immobilization of antygens on a substrate, the use of laser beam and a photodetector.

The smart chair described in [269] uses capacitive sensing to estimate respiratory rate, recognize the performed activity and detect inappropriate postures that can unleash chronic back pain.

The analysis of gait has also been related to pain [270]. A study during pregnancy and postpartum with pressure sensors located in instrumented foot insoles correlated the prevalence of low back pain with alterations in the gait: increased pressure in the hindfoot and excessive pronation [270].

Other approaches employ data fusion from different sensors. These methods are beneficial because they allow a complementary and redundant use of information that can provide a more

accurate and reliable result [271–273]. Data fusion can reduce uncertainty in the evaluation stage and reduce the risk of loss of information, noise or errors in measurements of a particular sensor. On the other hand, complementary information from multiple sensors allows access to environmental characteristics that would not be possible to perceive in a single sensor approach [271].

Data fusion at the processing level can be considered at two levels [54]: early fusion, by combining different categories into a single data stream; or late fusion, by integrating outputs of individual classifiers as inputs to the next level of classification.

Multiple approaches to data fusion have been proposed in the literature. In [274], audio, video, temperature, ECG, EMG, and skin conductance signals were used to better pain classification. Sickle cell disease patients experiencing chronic pain were evaluated with heart rate, accelerometry, gyroscopy, temperature and skin conductance sensors [275]. In [276] and in [49], EEG and fMRI data fusion provided an improvement in accuracy. Patients with sickle cell disease were evaluated in [277] using systolic and diastolic blood pressure, peripheral capillary oxygen saturation, respiratory rate, heart rate and temperature sensors. In [278], chronic pain using heart rate and blood pressure sensors was evaluated. Signals of skin conductance levels, EMG and ECG were fused in [51] to analyze the response to thermal pain induced in healthy patients.

The system described in [279] proposes a multisensory approach (accelerometry, photoplethysmography (PPG), body temperature and electrodermal activity) to detect a migraine event in a previous state in order to notify patients to take their medications with enough time. In [280], another multisensory approach was used, based on acceleration and electromyogram sensors, to assess low back pain.

Pain intensity induced thermally was recognized in [27] through the joint use of video and audio signals (respiration sounds and sporadic moans acquired by a headphone located in the nasolabial region).

Finally, and although it is not directly related to the detection of pain, but to its avoidance, it is worth mentioning a system for the recognition of the distribution and imbalances of the weight of backpacks, providing feedback to the user to avoid problems related to back pain [281].

8. Processing Algorithms and Computational Models for Pain Assessment

As the evaluation of chronic pain is a complex task that often poses a challenge due to the amount of data to be managed, especially in the case of facial expression images, MRI, and multisensory approaches combining different approaches described so far, in recent years numerous machine learning algorithms have been proposed [253,255,282]. In a generic way, pain assessment is usually performed through procedures in which the following main stages can be distinguished:

8.1. Preprocessing

A preprocessing is applied to physiological signals to remove unwanted artifacts [51]. This preprocessing can be based on band-pass filters to remove motion artifacts and drifts from the continuous level of the signal, associated with the low-frequency components [49,50], or noise, normally included in the high-frequency components [35,50]. The electromagnetic interference component associated with the power grid can be removed by a notch filter at 50/60 Hz.

8.2. Feature Extraction

In the feature extraction stage, an abstraction of the relevant information from larger data elements of the physiological signal is performed [51,251,275]. The objective of this stage is to improve the density of information. Data extraction may be related to characteristics of the signal in the temporal domain, but also in the frequency domain [49]. Some features may be amplitude, frequency, stationarity, linearity, variability, similarity, entropy, etc. [35,51]. The features can in turn be standardized to keep all parameters in a comparable range of values [35].

8.3. Feature Selection

To decrease the computational cost, the subset of features that provide meaningful information related to the pain classification problem is selected [51,275]. In this way, noisy or redundant features can be removed for the subsequent analysis [251]. The objective of this stage is to improve the quality of the information. Univariate feature selection (UFS) implies the selection of features through a machine learning algorithm that iteratively improves a quality criterion (correlation coefficient, analysis of variance (ANOVA) F-value, mean square error, etc.) to yield a desired threshold [35,51,251]. In contrast, in sequential forward selection (SFS), the final classifier is used in the selection of the best features [251,275]. Principal component analysis (PCA) can also be used to reduce the number of features [216,283].

8.4. Classification

After extracting features, the pain classification algorithm takes place, composed of a first stage for pain assessment model training and a second stage for the test [35,251]. A common approach is to use a cross-validation, in which one part of the data is used to train the model, and another part is dedicated to the test, both independent from each other to model a realistic error [251,275]. Data augmentation is also often required to increase the number of entries used in the training process [239]. Some of the models that have been used for pain assessment are the following (at the end of each model some successfully studies related to the evaluation of chronic pain are shown):

- K-nearest neighbors (kNN): In this method, the output is the class resulting in the classification, which is obtained by a majority of votes of its closest *k* neighbors: [277,284].
- Support vector machine (SVM): In SVM model the data are represented as points in a hyper-space of high dimension. The optimization process tries to find the hyper-plane that defines the separation surface between categories, maximizing the distance from all points: [35,46,51,54,258, 277,283–286].
- Multinomial logistic regression (MLR): MLR is a classification method that generalizes logistic regression (a binary regression model that uses the logistic function to model the binary dependence) with a multiclass problem. The MLR model will finally be the probability of assignment to a certain class, choosing the one that provides the greatest probability: [258,277].
- Bayesian networks (BN): BN is a graphic model that specifies a probability distribution in a set of random variables. It consists basically of two components: a probability distribution of the variables and a directed graph that shows the dependencies between variables: [258,284].
- Restricted Boltzmann machine (RBM): It is a graphic model with symmetric connections between observable and hidden variables to model a probability distribution. The model is called restricted because it has no connections between elements of the same layer, which favors the learning process: [278].
- Relevance vector machine (RVM) and Gaussian process (GP) models: RVM and GP are Bayesian
 extensions of the SVM algorithm, that allow assessing the pain scale based on the probability of
 belonging to a class: [244,253].
- Decision tree (DT): DT is a graph-shaped model that represents the decision points as ramifications and the desired prediction as end-nodes or leaves. It is therefore a rule-based model that induces solutions from the decision rules: [258,284,287–289].
- Random forest (RF): An RF is a set of DTs, where each DT has a subset of leaf nodes. No tree
 has all training data. This causes each tree to be trained with different data samples for the same
 problem. In this way, by combining their results some errors are compensated with others and a
 more generalized prediction is obtained: [50,290–292].

Models based on artificial neural networks (ANNs) have also been used in pain assessment because of their ability to learn, generalize problems and abstract essential characteristics from inputs with apparent irrelevant information [293]:

- Multilayer perceptron (MLP): The multilayer perceptron is an ANN formed by multiple layers of a basic unit or artificial neuron known as perceptron, with a forward directed flow of information from the input layer to the output layer, passing through the hidden layer/layers, and with a learning algorithm through backward propagation: [258,284,293,294]
- Deep learning (DL): DL is based on a division in multiple levels that correspond to different levels of abstraction for data expressed in matrix form: [295].
- Convolutional neural networks (CNNs): This ANN is an extension of MLP applied to two-dimensional matrices through a convolution operation [295], more suitable for computer vision tasks. In an MLP, the output of each neuron depends on all inputs. This structure does not allow the network to model patterns such as those that can be found in an image. CNN neurons are connected to a small number of inputs belonging to a small continuous region [293]. Convolutional kernels are used to extract multiple local characteristics and adjust to different patterns. This characteristic considerably reduces the number of parameters and allows a greater specificity in the recognition of characteristics: [52,239,293,295–300].
- Deep belief networks (DBNs): It is a type of DL with connections between layers but not between units within the same layer: [239].
- Recurrent neural networks (RNNs): This type of ANN represents a directed graph capable of abstracting temporal sequences of data, providing the ability to model dynamic behaviors: [239, 293].
- Mixed networks (MNs): They are hybrid networks formed by the interconnection of two or more different ANNs, such as CNN and RNN: [54,239].

Table 1 shows a summary of some of the results obtained with machine learning techniques in pain assessment studies (pain classification accuracy defined according to [284]).

Study	Pain etiology	Machine learning model	Sensor data	Pain classification accuracy
[239]	Shoulder pain	CNN,RNN	Facial expressions (computer vision)	75%
[255]	Chronic low back pain, osteoarthritis and fibromyalgia	SVM	fMRI	75%
[241]	Infant pain	SVM	Facial expressions (computer vision)	83.8%
[45]	Induced pain in healthy people	k-NN	fNIRS	92.1%
		SVN	fNIRS	91.2%
[290]	Musculoskeletal chronic pain	RF	Electronic health records	94%
[244]	Infant pain	RVM	Facial expressions (computer vision)	91%
[252]	Chronic low back pain	SVM	fMRI	92.5%
[258]	Fibromyalgia	DT	fMRI	76%
[48]	Induced pain in healthy people	K-NN	fNIRS	88.3%
[216]	Shoulder pain	SVM	EEG	84%
[298]	Multiple pain etiology	CNN	Facial expressions (computer vision)	93.3%
[283]	Pain after surgery	SVM	Facial expressions (computer vision)	87%
[286]	Pain after surgery	SVM	Skin conductance	77.7%
[299]	Shoulder pain	CNN	Facial expressions (computer vision)	98.5%
[300]	Infant pain	CNN	Facial expressions (computer vision)	88.3%
[49]	Induced pain in healthy people	SVM	EEG	78.2%
[50]	Induced pain in healthy people	RF	EEG	89.5%
[277]	Sickle cell disease pain	k-NN, SVM	Multisensor	68%
[278]	Multiple chronic pain	RBM	Heart rate, blood pressure	72%
[52]	Induced pain in healthy people	CNN	EEG	97.4%
[51]	Induced pain in healthy people	SVM	EMG, skin conductance, ECG	79.4%
[54]	Neck and shoulder pain	SVM	EMG	77%

Table 1. Processing algorithms examples for pain assessment.

9. Digital Resources for the Evaluation of Pain

Finally, this section compiles diverse digital resources (database, mobile apps, etc.) developed in the context of chronic pain detection [301,302].

The system proposed in [37,303] recognizes physical pain situations under rehabilitation using high-resolution video images for the capture of facial expressions, a system for the 3D acquisition of full-body movements (Animazoo IGS-190) and four EMG sensors (BTS FREEEMG 300) placed symmetrically on the upper trapezius fibers and lumbar para-spinal muscles. The acquired data are available for the scientific community through a web link [304]. This database has been used for the chronic lumbar pain recognition through the detection of behaviours related to pain, such us guiding actions or sudden movements [7]. It has also been used for the automatic detection of protecting

behaviours [305] or the classification of pain levels ("none", "high", "low") during rehabilitation process [306]. The same records were used in [11] to discriminate low-intensity from high-intensity pain in full-trunk flexion movements (94% success), and sit/stand activities (80% success).

Pain situations were detected in [307] using UNBC McMaster database [308], with a positive-trues rate of 82.4%. This database contains video sequences of spontaneous facial expressions for patients with shoulder pain. The same database has been used in other studies, like [20], which reported a system that recognizes four pain intensities, or [29], which calculated a continuous variable related to pain intensity. The system presented in [309] established the probability of future pain considering current pain state.

Another database for pain research is BioVid Heat Pain [42,43]. This database includes video images, EMG, ECG, and skin conductance records from 90 volunteers under normal situation and heat-induced pain. This database was used in [23] to develop a detection system that obtained a success of 80%.

SenseEmotion is a database generated by the Department of Psychosomatic and Psychotherapy Medicine of Ulm University [19]. It includes signals acquired with diverse sensors (skin conductance, ECG, EMG, respiratory rate, video recording of facial expressions, and audio recordings), which were obtained from volunteers undergoing a set of random thermally-induced pain experiments.

X-ITE Pain Database will provide sensory information from multiple sensors (video, audio, skin temperature, ECG, skin conductance, EMG, among others) obtained from subjects undergoing stimuli of different intensity (low, medium, and high), duration (5 s/1 min) and mode (electric/heat pain) [274].

Different applications have been proposed to be executed in smartphones. Pain Care [310] is a mobile app that assists patients with chronic pain in the management of their symptoms, drugs and communication with medics. Another example is the app described in [311], which performs an evaluation of pain using facial images acquired with a smartphone. Images are decomposed in vector subspaces with sub-images as characteristic vectors defining the image. The mobile app [3] allows the classification of pain level ("normal", "moderate", "severe") using information from different sensors (heart rate, skin conductance, respiratory rate and skin temperature). Another app has been used to assess the level of wrist movement, which was inversely related to the degree of wrist pain self-reported by the patient [312]. In [313], an app is described that uses machine learning on physical activity data and pain self-report documents to find routine behaviors and automatically generate physical activity recommendations. The app proposed in [314] is aimed at the automatic detection of pain by recognizing spontaneous facial expressions using Google's Face API. Figure 8 shows an example of facial landmarks extracted by Google's Face API that are employed in the app.



Figure 8. Example of landmarks extracted by Google's Face API according to [314].

In [315], the authors propose a virtual reality (VR) environment for the motivation of patients in therapies for chronic pain relief. Another example of how biomedical engineering and virtual reality (VR) can help chronic pain relief is shown in [316].

10. Future Trends and Conclusions

Pain assessment is critical in a wide range of disorders [20]. The prevalence of chronic pain is global [313]. This condition is closely related to the state of psychological well-being, and is often in close union with anxiety problems, depressive disorders and negative affectivity [313]. Subjective and self-reported account of patients, caregivers or medical staff are currently the common method to measure pain [21]. However, self-assessed reports have limitations due to their subjective nature that can lead to inadequate and insufficient therapy [20,27].

Although the objective measurement of pain using biomedical sensors has been raised as a promising approach to address this problem, the precise measurement of pain remains a challenge for researchers and healthcare professionals [28]. Objective pain assessment has been investigated from multiple approaches. HRV has been used for the evaluation of chronic pain of different etiology (headache, neck, shoulder, lower back, fibromyalgia, hip, gastrointestinal, burning mouth syndrome, etc.) [28,77,91–97,105,109]. Different studies have shown that people with chronic pain have decreased HRV compared to people without pain, especially in the high frequency component [77,95].

On the other hand, it is common that patients who feel chronic musculoskeletal pain are less physically active [123]. The anxiety produced by the feeling of pain causes patients to behave in a self-protective manner, avoiding the type of movement that causes the pain [11]. However, and despite all these conditions, the measurement of physical activity remains a challenge in the treatment, rehabilitation, and health promotion of patients with chronic non-oncological pain [131]. Physical activity monitoring devices have the potential to help clinicians in multiple novel pain applications, such as surgeon support tools to provide personalized post-operative care [317].

Additionally, a greater research effort is required to understand the mechanisms related to movement deficits in patients with pain, including monitoring processes through sensor technologies in routine clinical protocols [129]. More studies are needed to analyze the characteristics of movements during gait or posture and their correlation with pain. These studies can be used to design exercise programs for preventive treatment and pain rehabilitation [270].

Frequent physical activity can attenuate pain level in the case of chronic musculoskeletal pain, however, patients with pain tend to limit physical activity as a result of the psychological association established between pain and movement [313,318]. Self-reported pain and movement disability are directly related to the natural tendency to avoid pain [125]. Despite the research progress, more studies are needed that relate the kinematics of body movements to pain avoidance mechanisms [125]. To achieve this, an external feedback mechanism is essential to allow patients to learn and correct their movement habits.

On the other hand, the monitoring of physical activity may be affected by circumstances of personal mood, weather conditions, and other psycho-social factors [25]. Future studies should incorporate these aspects to provide more reliable measures of physical activity, such as variability or regularity in the development of activities, and symmetry of gait [25]. The effect of physical activity on pain is very broad, and new studies are necessary to incorporate an analysis of the frequency, intensity, time, and type of physical activity [174]. These studies should be evaluated by classifying patients into subgroups, according to their respective activity limitations.

Skin conductance has also been applied to the evaluation of pain, since there can be a rapid increase in its value as a result of an increase in the secretion of sweat glands, innervated by the sympathetic branch of the ANS.

Other systems used in pain assessment are based on EMG signals, since a high level of muscle tension is an indicator of pain [35,43]. Facial EMG is an alternative technique to assess pain perception

through facial muscle activity [319]. However, deeper studies are necessary to prove the correlation with pain.

Image processing and computer vision techniques have also been used in the recognition of pain [243] or the evaluation of pain intensity [244]. Facial movements associated with pain feeling can be identified with computer vision techniques [319]. This procedure has some limitations, such as the movement artifacts of subjects that must be addressed before the translation to clinical practice [319]. Other problems are the complexity of image-based systems and the cost of the devices. Important research challenges are posed to mitigate the aforementioned limitations. The processing of audio signals can also be useful by means of the detection of noise and groans related to pain [27].

Other parameters have also been used in the evaluation of pain, such as the monitoring of respiratory rate [43], since negative emotions generally induce an irregular respiratory pattern [78]. Blood pressure has also been a study parameter, which may be higher in pain situations [108]. The body temperature has also been considered, since in situations of pain the muscles are under tension, the blood vessels contract and, therefore, the temperature decreases [43]. Photoplethysmography can also be used to measure vasoconstriction, which may be related to pain states [43,266].

However, there is still no universal method that allows the objective evaluation of pain [22], its intensity or severity, although it may be critical in multiple conditions [28].

One of the major limitations hindering the objective evaluation of pain is the lack of studies with clinically valid data [320]. It is necessary to incorporate biomarkers in the evaluation studies that allow to establish correspondences with a reference standard, such as yclooxygenase-2 or the inducible nitric oxide synthase [320]. Further research is necessary through randomized clinical trials on different populations and considering different therapeutic modalities to confirm the results of previous studies and provide more evidence on the possibility of objectively assessing chronic pain with sufficient precision to be useful toward a more personalized, proactive and preventive medicine [321].

The measurement of pain in non-communicative patients is a research challenge [231,319], which extends from infants and children [322] to adults in the post-operative period, in intensive care units or with cognitive impairment. An application of great interest is the closed-loop peri-operative monitoring with the purpose of automatically controlling the administration of anesthesia [319].

People who work in a closed environment and sit in a chair sometimes maintain inappropriate postures that can lead to damage and affections in the spine [323]. Some proposals based on accelerometers and gyroscopes have been made to detect incorrect postures and provide tactile feedback through vibrations to notify the user of the need to correct the posture. Low back pain has a great impact on work productivity worldwide, despite years of research in this area [280]. Understanding the mechanisms involved in the injuries that cause pain can be useful to improve prevention and treatment in occupational risk protocols. New solutions based on feedback mechanisms to users would be highly recommended, which can alert the user when they remain in a position for a long time or when they make inappropriate movements [280]. The feeling of security that could be assumed by the subject would improve their quality of life and the well-being of the worker and would have a positive health impact since harmful positions would be avoided.

Rehabilitation is essential in the control of multiple musculoskeletal conditions [132] related to pain. Rehabilitation treatment is largely based on the performance of a series of exercises in the home environment, but on many occasions patients do not adequately comply with the prescribed exercise and the degree of compliance with the rehabilitation program is not known. There is a need for technological tools that allow the supervision of rehabilitation therapies in patients' homes, positively influencing the motivation and control of patient adherence [132]. Physical rehabilitation must be adapted by physiotherapists according to the movement abilities of patients [124]. In this sense, technology can provide fundamental support in physical rehabilitation, to the extent that it would allow the evaluation of patient movement patterns to provide such personalized follow-up [124].

In the rehabilitation process, the ability to provide visual and acoustic feedback to the patient to promote their motivation in compliance with the prescribed rehabilitation program should be investigated [129]. The aspects related to the retro-feedback of patients remain unexplored. It would be of great interest to reinforce research into automated technological solutions that enable positive beliefs, stimulation and self-efficacy of physical activity, since the overload of medical staff prevents the patient from being treated in a holistic and continuous way [313]. Future studies should also analyze the relationships between the improvement produced in pain and the mechanisms of intervention and feedback to patients [280]. In these studies, a multi-causal approach should be considered in order to distinguish between the benefits directly caused by the reduction of pain from other additional psychosocial factors that can cause similar or placebo effects [280].

The non-intrusiveness of the measurements carried out by the sensor devices during the subject's daily life is another challenge to be approached in future research [169]. The same concerns apply to the development of pain assessment technologies through the miniaturization of the devices to achieve true continuous monitoring 24 hours a day [266].

Although many studies have evaluated pain by observing individual physiological parameters, future studies should employ a multisensory point of view to allow a deeper understanding of the causes and effects of pain [280]. However, a balance must be established between the complexity and cost of the systems with respect to the benefits provided in precision. Future studies should explore the cost-effectiveness of the systems and the possibilities of inclusion in the clinical guidelines for patient healthcare [280]. These research studies should focus on randomized clinical trials over larger populations, also incorporating an economic evaluation [280].

The incorporation of the new technologies in pain assessment systems is another area of great development potential, either supported by smartphones or smart watches, to the extent that many of these systems already incorporate means for the evaluation of physiological signals such as heart rate or physical activity in real time [301]. Smartphones can be used to collect the results of pain self-assessment reports more efficiently [312]. In turn, the sensors incorporated in smartphones can be used to gather clinical information of relevance, which may be of particular interest in the evaluation of pain [312]. This approach can be expanded with the incorporation of social networks [312]. Internet-of-Things (IoT) is a recent area of study of great popularity and objective pain assessment can be benefited by this ubiquitous monitoring approach [302].

Artificial neural networks (ANNs) are currently considered a powerful method for data analysis and pattern recognition [324]. In particular, the application of so-called deep learning neural networks have grown with great intensity in recent years [325]. The design of the architecture of these networks remains an open area of study and their application to the objective evaluation of pain is clearly justified when physiological signals of different etiology are evaluated [324]. The objective measurement of pain through biomedical sensors is not ready for clinical use, hence there are enormous research and innovation opportunities in the field of the application of sensor technologies combined with artificial intelligence and machine learning techniques [319].

A greater research and development effort in health technologies is necessary, both in private and public assistance models, including rural areas and taking into account the vision and needs of the pain-affected patients [317]. There is a need for new systems and applications for chronic pain self-control, incorporating the needs and context of the user, which would allow improving accessibility to treatment, and reducing economic costs and waiting times [314]. However, current pain management developments have not been designed with integrated features that effectively address the multidimensional nature of pain.

Acute and chronic pain are two different clinical conditions of pain [326], however, many studies do not discriminate between acute and chronic pain [327], partly because among the challenges to both researchers and clinicians is the understanding and distinction between acute and chronic pain [328]. Although many studies address acute nociceptive pain, sometimes performed by phasic pain stimuli (short duration), the number of studies based on tonic pain stimuli is growing as an approach to chronic pain [41,42,44,51,52]. This validation approach represents an objective way for assessing chronic pain, leaving aside all the subjectivity that may be related to the perception of pain by the chronic patient.

Chronic pain is in general more challenging than acute pain, but also has a larger impact on society, so that research into chronic pain assessment methods of different etiology should be enhanced.

More studies are needed to analyze the effects of treatments and interventions on pain, as well as their functional manifestations, which would allow to optimize physical and clinical interventions on patients suffering from pain [24]. For this purpose, the integration of the information from these monitoring systems with the health systems in a standardized manner is necessary and remains a challenge for researchers and developers [301].

In conclusion, objective sensor-based pain monitoring technologies have the potential to help improving the patient's quality of life [312], to understand better the pain biopsychosocial processes and may be of interest in the study of other related conditions [172].

Author Contributions: All authors have participated actively in the development of this work. All authors have read and agreed to the published version of the manuscript. Conceptualization, D.N.-H., J.R.-T., and L.M.R.; methodology, D.N.-H., J.R.-T., and L.M.R.; formal analysis, D.N.-H., J.R.-T., and L.M.R.; investigation, D.N.-H., J.R.-T., and L.M.R.; writing—original draft preparation, D.N.-H.; writing—review and editing, D.N.-H. and J.R.-T.; supervision, D.N.-H., J.R.-T., and L.M.R.; project administration, J.R.-T. and L.M.R.; funding acquisition, J.R.-T. and L.M.R. All authors have read and agreed to the published version of the manuscript.

Funding: This work was supported in part by the "Fondo de Investigaciones Sanitarias" (Instituto de Salud Carlos III, Spain) under Grants PI15/00306, in part by the "Fundación Progreso y Salud" (Government of Andalucía, Spain) under Grant PIN-0394-2017, in part by "Fundación Mutua Madrileña" under grant VÍA-RENAL, and in part by the "Eurostars Programme", European Commission, under Grant "FRAIL".

Conflicts of Interest: The authors declare no conflict of interest. The founding sponsors had no role in the design of the study; in the collection, analyses, or interpretation of data; in the writing of the manuscript, and in the decision to publish the results.

Abbreviations

The following abbreviations are used in this manuscript:

ANN	Artificial Neural Networks
ANOVA	Analysis of Variance
ANS	Autonomic Nervous System
API	Application Programming Interface
BIS	Bioimpedance Spectrometry
BN	Bayesian Networks
BPAQ	Baecke Physical Activity Questionnaire
BSN	Body Sensor Networks
CNN	Convolutional Neural Network
DBN	Deep Belief Networks
DCT	Discrete Cosine Transform
DL	Deep Learning
DT	Decision Tree
ECG	Electrocardiagram
EEG	Electroencephalogram
ELISA	Enzyme-Linked Immunosorbent Assay
EMG	Electromyogram
EMI	Electromagnetic Interference
FACS	Facial Action Coding System
FFT	Fast Fourier Transform
fMRI	Functional Magnetic Resonance Imaging
fNIRS	Functional Near-Infrared Spectroscopy
gLASSO	Sparse Graphical Model
GP	Gaussian Process
HbO	Oxygenated Hemoglobin
HbR	Deoxygenated Hemoglobin
HF	High-Frequency

HOG	Histogram of Oriented Gradients
HRV	Heart Rate Variability
Iot	Internet of Things
k-NN	k-Nearest Neighbors
LBP	Local Binary Patterns
LF	Low-Frequency
MLP	Multilayer Perceptron
MLR	Multinomial Logistic Regression
MN	Mixed Networks
MORE	Mindfulness-Oriented Recovery Enhancement
MPQ	Mcgill Pain Questionnaire
MRI	Magnetic Resonance Imaging
MVPA	Multivariate Pattern Analyses
PCA	Principal Component Analysis
PPI	Present Pain Intensity
PRI	Pain Rating Index
PSD	Power Espectral Density
PPG	Photoplethysmograph
RBM	Restricted Boltzmann Machine
RF	Random Forest
RM	Regression Models
RNN	Recurrent Neural Networks
ROI	Regions of Interest
RVM	Relevance Vector Machine
Semg	Surface Electromyogram
SFS	Sequential Forward Selection
SIFT	Scale-Invariant Feature Transform
SURF	Speeded Up Robust Features
SVM	Support Vector Machine
UFS	Univariate Feature Selection
VAS	Visual Analog Scale
VLF	Very-Low-Frequency
VR	Virtual Reality
WT	Wavelet Transform

References

- 1. Werner, P.; Lopez-Martinez, D.; Walter, S.; Al-Hamadi, A.; Gruss, S.; Picard, R. Automatic Recognition Methods Supporting Pain Assessment: A Survey. *IEEE Trans. Affect. Comput.* **2019**, in press.
- 2. Nesse, R.; Schulkin, J. An evolutionary medicine perspective on pain and its disorders. *Philos. Trans. R. Soc. B Biol. Sci.* **2019**, *374*, 1785.
- Rajesh, M.; Muthu, J.; Suseela., G. IPainRelief -A pain assessment and management app for a smart phone implementing sensors and soft computing tools. In Proceedings of the 2013 International Conference on Information Communication and Embedded Systems (ICICES 2013), Chennai, India, 21–22 February 2013; pp. 434–441.
- 4. Kato, J.; Agalave, N.; Svensson, C. Pattern recognition receptors in chronic pain: Mechanisms and therapeutic implications. *Eur. J. Pharmacol.* **2016**, *788*, 261–273.
- 5. Park, Y.; Kang, S.; Kim, Y. Successful application of the erector spinae plane block for the management of zoster-associated pain. *J. Clin. Anesth.* **2020**, *60*, 70–71.
- 6. Raffaeli, W.; Arnaudo, E. Pain as a disease: An overview. J. Pain Res. 2017, 10, 2003–2008.
- Qin, R.; Meng, H.; Li, M. Continuous pain related behavior recognition from muscle activity and body movements. In Proceedings of the 2016 12th International Conference on Natural Computation, Fuzzy Systems and Knowledge Discovery (ICNC-FSKD 2016), Changsha, China, 13–15 August 2016; pp. 1710–1715.

- 8. Mostoufi, S.; Afari, N.; Ahumada, S.; Reis, V.; Wetherell, J. Health and distress predictors of heart rate variability in fibromyalgia and other forms of chronic pain. *J. Psychosom. Res.* **2012**, *72*, 39–44.
- 9. Lee, I.S.; Necka, E.; Atlas, L. Distinguishing pain from nociception, salience, and arousal: How autonomic nervous system activity can improve neuroimaging tests of specificity. *NeuroImage* **2020**, *204*, 116254.
- Lindheimer, J.; Stegner, A.; Ellingson-Sayen, L.; Van Riper, S.; Dougherty, R.; Falvo, M.; Cook, D. Influence of pain anticipation on brain activity and pain perception in Gulf War Veterans with chronic musculoskeletal pain. *Psychophysiology* 2019, *56*, doi:10.1111/psyp.13452.
- Olugbade, T.; Bianchi-Berthouze, N.; Marquardt, N.; Williams, A. Pain level recognition using kinematics and muscle activity for physical rehabilitation in chronic pain. In Proceedings of the 2015 International Conference on Affective Computing and Intelligent Interaction (ACII 2015), Xi'an, China, 21–24 September 2015; pp. 243–249.
- 12. Dsingh, A.; Kaur, A. Role of postural control exercises and pelvic floor strengthening exercises on chronic low back pain of women with sitting jobs. *Adv. Intell. Syst. Comput.* **2020**, *955*, 775–782.
- 13. Häuser, W.; Clauw, D.; Fitzcharles, M.A. Fibromyalgia as a chronic primary pain syndrome: issues to discuss. *Pain* **2019**, *160*, 2651–2652.
- Pourahmadi, M.; Mohseni-Bandpei, M.; Keshtkar, A.; Koes, B.; Fernández, C.; Dommerholt, J.; Bahramian, M. Effectiveness of dry needling for improving pain and disability in adults with tension-type, cervicogenic, or migraine headaches: Protocol for a systematic review. *Chiropr. Man. Ther.* 2019, 27, 43.
- Said Ahmed, M.; Boles Saweeres, E.; Abdelkader, N.; Abdelmajeed, S.; Fares, A. Improved pain and function in knee osteoarthritis with dexamethasone phonophoresis: A randomized controlled trial. *Indian J. Orthop.* 2019, 53, 700–707.
- Eather, A.; Kenardy, J.; Healy, K.; Haynes, M.; Sterling, M. How are pain and traumatic stress symptoms related in acute whiplash-associated disorders? An investigation of the role of pain-related fear in a daily diary study. *Pain* 2019, *160*, 1954–1966.
- 17. Sharp, A.; Baecker, A.; Shen, E.; Redberg, R.; Lee, M.S.; Ferencik, M.; Natsui, S.; Zheng, C.; Kawatkar, A.; Gould, M.; et al. Effect of a HEART Care Pathway on Chest Pain Management Within an Integrated Health System. *Ann. Emerg. Med.* **2019**, *74*, 171–180.
- 18. Nezhat, C.; Vang, N.; Tanaka, P.; Nezhat, C. Optimal Management of Endometriosis and Pain. *Obstet. Gynecol.* **2019**, *134*, 834–839.
- Velana, M.; Gruss, S.; Layher, G.; Thiam, P.; Zhang, Y.; Schork, D.; Kessler, V.; Meudt, S.; Neumann, H.; Kim, J.; et al. The senseemotion database: A multimodal database for the development and systematic validation of an automatic pain-and emotion-recognition system. *Lect. Notes Comput. Sci.* 2017, 10183 *LNAI*, 127–139.
- 20. Hammal, Z.; Cohn, J. Automatic detection of pain intensity. In Proceedings of the ACM International Conference on Multimodal Interaction (ICMI'12), Santa Monica, CA, USA, 22–26 October 2012; pp. 47–52.
- 21. Robinson-Papp, J.; George, M.; Dorfman, D.; Simpson, D. Barriers to Chronic Pain Measurement: A Qualitative Study of Patient Perspectives. *Pain Med.* **2015**, *16*, 1256–1264.
- 22. Singh, S.; Roy, A. Assessment of heart rate variability in the patients suffering with chronic pain of musculoskeletal origin. *Natl. J. Physiol. Pharm. Pharmacol.* **2017**, *7*, 712–718.
- Walter, S.; Gruss, S.; Traue, H.; Werner, P.; Al-Hamadi, A.; Kachele, M.; Schwenker, F.; Andrade, A.; Moreira, G. Data fusion for automated pain recognition. In Proceedings of the 2015 9th International Conference on Pervasive Computing Technologies for Healthcare (PervasiveHealth 2015), Istanbul, Turkey, 20–23 May 2015; pp. 261–264.
- 24. Forechi, L.; Silveira-Nunes, G.; Barbosa, M.; Barbosa, E.; Santos, D.; Vieira, E.; Barbosa, A. Pain, balance, grip strength and gait parameters of older adults with and without post-chikungunya chronic arthralgia. *Trop. Med. Int. Health* **2018**, *23*, 1394–1400.
- 25. Yen, T.; Mohler, J.; Dohm, M.; Laksari, K.; Najafi, B.; Toosizadeh, N. The effect of pain relief on daily physical activity: In-home objective physical activity assessment in chronic low back pain patients after paravertebral spinal block. *Sensors* **2018**, *18*, 3048.
- 26. Katz, J.; Melzack, R. Measurement of pain. Surg. Clin. N. Am. 1999, 79, 231-252.
- 27. Thiam, P.; Kessler, V.; Walter, S.; Palm, G.; Schwenker, F. Audio-visual recognition of pain intensity. *Lect. Notes Comput. Sci.* **2017**, *10183 LNAI*, 110–126.

- 28. Davydov, D.; Naliboff, B.; Shahabi, L.; Shapiro, D. Asymmetries in reciprocal baroreflex mechanisms and chronic pain severity: Focusing on irritable bowel syndrome. *Neurogastroenterol. Motil.* **2017**, 30, e13186.
- 29. Kaltwang, S.; Rudovic, O.; Pantic, M. Continuous pain intensity estimation from facial expressions. *Lect. Notes Comput. Sci.* **2012**, 7432 *LNCS*, 368–377.
- Kutluk, A.; Hirano, H.; Nakamura, R.; Saeki, N.; Yoshizumi, M.; Kawamoto, M.; Hamit, M.; Tsuji, T. Assessment of pain with mechanical nociceptive stimuli by the change of arterial wall impedance. In Proceedings of the 2012 5th International Conference on Biomedical Engineering and Informatics (BMEI 2012), Chongqing, China, 16–18 October 2012; pp. 451–454.
- Florea, C.; Florea, L.; Vertan, C. Learning pain from emotion: Transferred HoT data representation for pain intensity estimation. *Lect. Notes Comput. Sci.* 2015, 8927, 778–790.
- 32. Werner, P.; Al-Hamadi, A.; Niese, R.; Walter, S.; Gruss, S.; Traue, H. Towards pain monitoring: Facial expression, head pose, a new database, an automatic system and remaining challenges. In Proceedings of the British Machine Vision Conference, Bristol, UK, 9–13 September 2013; pp. 119.1–119.13.
- Amirian, M.; Kächele, M.; Schwenker, F. Using radial basis function neural networks for continuous and discrete pain estimation from bio-physiological signals. *Lect. Notes Comput. Sci.* 2016, 9896 LNAI, 269–284.
- 34. Chu, Y.; Zhao, X.; Yaof, J.; Zhao, Y.; Wu, Z. Physiological signals based quantitative evaluation method of the pain. *IFAC Proc. Vol.* **2014**, *19*, 2981–2986.
- 35. Gruss, S.; Treister, R.; Werner, P.; Traue, H.; Crawcour, S.; Andrade, A.; Walter, S. Pain intensity recognition rates via biopotential feature patterns with support vector machines. *PLoS ONE* **2015**, *10*, e0140330.
- 36. Kächele, M.; Thiam, P.; Amirian, M.; Schwenker, F.; Palm, G. Methods for Person-Centered Continuous Pain Intensity Assessment from Bio-Physiological Channels. *IEEE J. Sel. Top. Signal Process.* **2016**, *10*, 854–864.
- Aung, M.; Kaltwang, S.; Romera-Paredes, B.; Martinez, B.; Singh, A.; Cella, M.; Valstar, M.; Meng, H.; Kemp, A.; Shafizadeh, M.; et al. The automatic detection of chronic pain-related expression: Requirements, challenges and the multimodal EmoPain dataset. *IEEE Trans. Affect. Comput.* 2016, *7*, 435–451.
- 38. Kächele, M.; Thiam, P.; Amirian, M.; Werner, P.; Walter, S.; Schwenker, F.; Palm, G. Multimodal data fusion for person-independent, continuous estimation of pain intensity. *Commun. Comput. Inf. Sci.* 2015, *517*, 275–285.
- 39. Kächele, M.; Werner, P.; Al-Hamadi, A.; Palm, G.; Walter, S.; Schwenker, F. Bio-visual fusion for person-independent recognition of pain intensity. *Lect. Notes Comput. Sci.* **2015**, *9132*, 220–230.
- Werner, P.; Al-Hamadi, A.; Niese, R.; Walter, S.; Gruss, S.; Traue, H. Automatic pain recognition from video and biomedical signals. In Proceedings of the International Conference on Pattern Recognition, Stockholm, Sweden, 24–28 August 2014; pp. 4582–4587.
- 41. You, D.; Meagher, M. Association Between Borderline Personality Features and Temporal Summation of Second Pain: A Cross-Sectional Study. *Behav. Med.* **2017**, *43*, 208–217.
- 42. Institute for Information Technology and Communications, Germany. BioVid Heat Pain Database. Available online: http://www.iikt.ovgu.de/BioVid.html (accessed on 7 January 2020).
- 43. Walter, S.; Gruss, S.; Ehleiter, H.; Tan, J.; Traue, H.; Crawcour, S.; Werner, P.; Al-Hamadi, A.; Andrade, A.; Da Silva, G. The BioVid heat pain database: Data for the advancement and systematic validation of an automated pain recognition. In Proceeding of the 2013 IEEE International Conference on Cybernetics (CYBCONF 2013), Lausanne, Switzerland, 13–15 Junuary 2013; pp. 128–131.
- 44. Cruz-Almeida, Y.; Fillingim, R.; Riley, J.; Woods, A.; Porges, E.; Cohen, R.; Cole, J. Chronic pain is associated with a brain aging biomarker in community-dwelling older adults. *Pain* **2019**, *160*, 1119–1130.
- 45. Fernandez Rojas, R.; Huang, X.; Ou, K.L. Toward a functional near-infrared spectroscopy-based monitoring of pain assessment for nonverbal patients. *J. Biomed. Opt.* **2017**, *22*, 106013.
- 46. Gram, M.; Erlenwein, J.; Petzke, F.; Falla, D.; Przemeck, M.; Emons, M.; Reuster, M.; Olesen, S.; Drewes, A. Prediction of postoperative opioid analgesia using clinical-experimental parameters and electroencephalography. *Eur. J. Pain* **2017**, *21*, 264–277.
- Lindquist, M.; Krishnan, A.; López-Solá, M.; Jepma, M.; Woo, C.W.; Koban, L.; Chang, L.; Reynolds Losin, E.; Eisenbarth, H.; Ashar, Y.; Delk, E.; Wager, T.; Roy, M.; Atlas, L.; Reynolds Losin, E.; Krishnan, A.; Schmidt, L.; Schmidt, L. Group-regularized individual prediction: theory and application to pain. *NeuroImage* 2017, 145, 274–287.
- 48. Rojas, R.; Huang, X.; Romero, J.; Ou, K.L. fNIRS Approach to Pain Assessment for Non-verbal Patients. *Lect. Notes Comput. Sci.* **2017**, *10637 LNCS*, 778–787.

- 49. Tu, Y.; Tan, A.; Bai, Y.; Sam Hung, Y.; Zhang, Z. Decoding subjective intensity of nociceptive pain from pre-stimulus and post-stimulus brain activities. *Front. Comput. Neurosci.* **2016**, *10*, 32.
- 50. Vijayakumar, V.; Case, M.; Shirinpour, S.; He, B. Quantifying and Characterizing Tonic Thermal Pain Across Subjects From EEG Data Using Random Forest Models. *IEEE Trans. Biomed. Eng.* **2017**, *64*, 2988–2996.
- 51. Campbell, E.; Phinyomark, A.; Scheme, E. Feature extraction and selection for pain recognition using peripheral physiological signals. *Front. Neurosci.* **2019**, *13*, 437.
- 52. Yu, M.; Sun, Y.; Zhu, B.; Zhu, L.; Lin, Y.; Tang, X.; Guo, Y.; Sun, G.; Dong, M. Diverse frequency band-based convolutional neural networks for tonic cold pain assessment using EEG. *Neurocomputing* **2019**, doi:10.1016/j.neucom.2019.10.023.
- 53. Huishi Zhang, C.; Sohrabpour, A.; Lu, Y.; He, B. Spectral and spatial changes of brain rhythmic activity in response to the sustained thermal pain stimulation. *Hum. Brain Mapp.* **2016**, *37*, 2976–2991.
- 54. Hermens, H.; op den Akker, H.; Tabak, M.; Wijsman, J.; Vollenbroek, M. Personalized Coaching Systems to support healthy behavior in people with chronic conditions. *J. Electromyogr. Kinesiol.* **2014**, *24*, 815–826.
- 55. Huber, M.; Bartling, J.; Pachur, D.; Woikowsky-Biedau, S.; Lautenbacher, S. EEG responses to tonic heat pain. *Exp. Brain Res.* **2006**, *173*, 14–24.
- Chen, C.; Wang, Z.; Li, W.; Chen, H.; Mei, Z.; Yuan, W.; Tao, L.; Zhao, Y.; Huang, G.; Mei, Y.; Cao, Z.; Wang, R.; Chen, W. Novel flexible material-based unobtrusive and wearable body sensor networks for vital sign monitoring. *IEEE Sens. J.* 2019, 19, 8502–8513.
- 57. Niu, S.; Matsuhisa, N.; Beker, L.; Li, J.; Wang, S.; Wang, J.; Jiang, Y.; Yan, X.; Yun, Y.; Burnett, W.; et al. A wireless body area sensor network based on stretchable passive tags. *Nat. Electron.* **2019**, *2*, 361–368.
- 58. Lin, K.; Li, Y.; Sun, J.; Zhou, D.; Zhang, Q. Multi-sensor fusion for body sensor network in medical human–robot interaction scenario. *Inf. Fusion* **2020**, *57*, 15–26.
- 59. Wang, M.; Wang, X.; Yang, L.; Deng, X.; Yi, L. Multi-sensor fusion based intelligent sensor relocation for health and safety monitoring in BSNs. *Inf. Fusion* **2020**, *54*, 61–71.
- 60. Rachana, P.; Prasad, D. Green Communication in Wireless Body Sensor Network—A Review. *Lect. Notes Electr. Eng.* **2020**, *614*, 55–67.
- 61. Tarannum, S.; Farheen, S. Wireless Sensor Networks for Healthcare Monitoring: A Review. *Lect. Notes Netw. Syst.* **2020**, *98*, 669–676.
- 62. Harbouche, A.; Djedi, N.; Erradi, M.; Ben-Othman, J.; Kobbane, A. Model driven flexible design of a wireless body sensor network for health monitoring. *Comput. Netw.* **2017**, *129*, 548–571.
- 63. Kwasnicki, R.; Yang, G.Z. Clinical Applications of Body Sensor Networks. In *Handbook of Biomedical Telemetry*; Nikita, K.S., Ed.; John Wiley & Sons, Inc.: Hoboken, NJ, USA, 2014; pp. 479–504.
- 64. Ausín, J.; Duque-Carrillo, J.; Ramos, J.; Torelli, G. From handheld devices to near-invisible sensors: The road to pervasive e-health. *Smart Sens. Meas. Instrum.* **2013**, *2*, 135–156.
- 65. Bachmann, C.; Ashouei, M.; Pop, V.; Vidojkovic, M.; Groot, H.; Gyselinckx, B. Low-power wireless sensor nodes for ubiquitous long-term biomedical signal monitoring. *IEEE Commun. Mag.* **2012**, *50*, 20–27.
- 66. Yoo, J.; Yoo, H.J. Emerging low energy Wearable Body Sensor Networks using patch sensors for continuous healthcare applications. In Proceeding of the 2010 Annual International Conference of the IEEE Engineering in Medicine and Biology Society (EMBC'10), Buenos Aires, Argentina, 31 August–4 September 2010; pp. 6381–6384.
- Fortino, G.; Guerrieri, A.; Bellifemine, F.; Giannantonio, R. Platform-independent development of collaborative Wireless Body Sensor Network applications: SPINE2. In Proceeding of the Conference Proceedings - IEEE International Conference on Systems, Man and Cybernetics, San Antonio, TX, USA, 11–14 October 2009; pp. 3144–3150.
- Chua, M. Design and development of an integrated health (i-health) monitoring watch. In Proceeding of the SII 2016—2016 IEEE/SICE International Symposium on System Integration, Sapporo, Japan, 13–15 December 2016; pp. 296–300.
- Konijnenburg, M.; Stanzione, S.; Yan, L.; Jee, D.W.; Pettine, J.; Van Wegberg, R.; Kim, H.; Van Liempd, C.; Fish, R.; Schuessler, J.; et al. A multi(bio)sensor acquisition system with integrated processor, power management, 8 × 8 LED drivers, and simultaneously synchronized ECG, BIO-Z, GSR, and ywo PPG readouts. *IEEE J. Solid-State Circuits* 2016, 51, 2584–2595. doi:10.1109/JSSC.2016.2605660.
- 70. Fortino, G.; Parisi, D.; Pirrone, V.; Di Fatta, G. BodyCloud: A SaaS approach for community Body Sensor Networks. *Future Gener. Comput. Syst.* **2014**, *35*, 62–79.

- 71. Zhang, Y.T.; Zheng, Y.L.; Lin, W.H.; Zhang, H.Y.; Zhou, X.L. Challenges and opportunities in cardiovascular health informatics. *IEEE Trans. Biomed. Eng.* **2013**, *60*, 633–642.
- 72. Caldeira, J.; Rodrigues, J.; Lorenz, P. Toward ubiquitous mobility solutions for body sensor networks on healthcare. *IEEE Commun. Mag.* **2012**, *50*, 108–115.
- 73. Azar, J.; Habib, C.; Darazi, R.; Makhoul, A.; Demerjian, J. Using Adaptive Sampling and DWT Lifting Scheme for Efficient Data Reduction in Wireless Body Sensor Networks. In Proceeding of the International Conference on Wireless and Mobile Computing, Networking and Communications, Limassol, Cyprus, 15–17 October 2018.
- 74. Chatterjee, S.; Chatterjee, S.; Choudhury, S.; Basak, S.; Dey, S.; Sain, S.; Ghosal, K.; Dalmia, N.; Sircar, S. Internet of things and body area network-an integrated future. In Proceeding of the 2017 IEEE 8th Annual Ubiquitous Computing, Electronics and Mobile Communication Conference (UEMCON 2017), New York, NY, USA, 19–21 October 2017; pp. 396–400.
- 75. Dey, N.; Ashour, A.; Shi, F.; Fong, S.; Sherratt, R. Developing residential wireless sensor networks for ECG healthcare monitoring. *IEEE Trans. Consum. Electron.* **2017**, *63*, 442–449.
- 76. Lautner, P. Aristotle on the intentional nature of emotions. Croat. J. Philos. 2012, 12, 221–237.
- 77. Koenig, J.; Thayer, J.; Falvay, D.; Clamor, A.; Wagner, J.; Jarczok, M.; Ellis, R.; Weber, C. Pneumogastric (vagus) nerve activity indexed by heart rate variability in chronic pain patients compared to healthy controls: A systematic review and meta-analysis. *Pain Phys.* **2016**, *19*, E55–E78.
- 78. Kim, J.; André, E. Emotion recognition based on physiological changes in music listening. *IEEE Trans. Pattern Anal. Mach. Intell.* **2008**, *30*, 2067–2083.
- 79. Gravina, R.; Fortino, G. Automatic Methods for the Detection of Accelerative Cardiac Defense Response. *IEEE Trans. Affect. Comput.* **2016**, *7*, 286–298.
- 80. Prakash, E. Sympathovagal balance from heart rate variability: An obituary, but what is sympathovagal balance? *Exp. Physiol.* **2012**, *97*, 1140–1140.
- 81. Van Den Houte, M.; Van Oudenhove, L.; Van Diest, I.; Bogaerts, K.; Persoons, P.; De Bie, J.; Van den Bergh, O. Negative affectivity, depression, and resting heart rate variability (HRV) as possible moderators of endogenous pain modulation in functional somatic syndromes. *Front. Psychol.* **2018**, *9*, 275.
- 82. Naranjo-Hernández, D.; Roa, L.; Reina-Tosina, J.; Barbarov-Rostan, G.; Galdámez-Cruz, O. Smart Device for the Determination of Heart Rate Variability in Real Time. *J. Sens.* **2017**, 2017, 1–11.
- 83. Chen, X.; Huang, Y.Y.; Yun, F.; Chen, T.J.; Li, J. Effect of changes in sympathovagal balance on the accuracy of heart rate variability obtained from photoplethysmography. *Exp. Therap. Med.* **2015**, *10*, 2311–2318.
- 84. Stein, P.; Domitrovich, P.; Huikuri, H.; Kleiger, R. Traditional and nonlinear heart rate variability are each independently associated with mortality after myocardial infarction. *J. Cardiovasc. Electrophysiol.* **2005**, *16*, 13–20.
- 85. Chou, Y.; Zhang, R.; Feng, Y.; Lu, M.; Lu, Z.; Xu, B. A real-time analysis method for pulse rate variability based on improved basic scale entropy. *J. Healthc. Eng.* **2017**, *2017*, 7406896.
- 86. Malik, M. Heart rate variability: Standards of measurement, physiological interpretation, and clinical use. *Circulation* **1996**, *93*, 1043–1065.
- 87. Chuang, C.C.; Ye, J.J.; Lin, W.C.; Lee, K.T.; Tai, Y.T. Photoplethysmography variability as an alternative approach to obtain heart rate variability information in chronic pain patient. *J. Clin. Monit. Comput.* **2015**, *29*, 801–806.
- Hadase, M.; Azuma, A.; Zen, K.; Asada, S.; Kawasaki, T.; Kamitani, T.; Kawasaki, S.; Sugihara, H.; Matsubara, H. Very Low Frequency Power of Heart Rate Variability is a Powerful Predictor of Clinical Prognosis in Patients with Congestive Heart Failure. *Circ. J.* 2004, *68*, 343–347.
- 89. Kudrynski, K.; Strumillo, P. Real-time estimation of the spectral parameters of Heart Rate Variability. *Biocybern. Biomed. Eng.* **2015**, *35*, 304–316.
- 90. Bunsawat, K.; Goulopoulou, S.; Collier, S.; Figueroa, A.; Pitetti, K.; Baynard, T. Normal HR with tilt, yet autonomic dysfunction in persons with down syndrome. *Med. Sci. Sports Exerc.* **2015**, *47*, 250–256.
- 91. Koenig, J.; Loerbroks, A.; Jarczok, M.; Fischer, J.; Thayer, J. Chronic pain and heart rate variability in a cross-sectional occupational sample evidence for impaired vagal control. *Clin. J. Pain* **2016**, *32*, 218–225.
- Cohen, H.; Neumann, L.; Shore, M.; Amir, M.; Cassuto, Y.; Buskila, D. Autonomic dysfunction in patients with fibromyalgia: Application of power spectral analysis of heart rate variability. *Semin. Arthritis Rheum.* 2000, 29, 217–227.

- 93. Maixner, W.; Greenspan, J.; Dubner, R.; Bair, E.; Mulkey, F.; Miller, V.; Knott, C.; Slade, G.; Ohrbach, R.; Diatchenko, L.; Fillingim, R. Potential autonomic risk factors for chronic TMD: Descriptive data and empirically identified domains from the OPPERA case-control study. J. Pain 2011, 12, T75–T91.
- 94. Staud, R. Heart rate variability as a biomarker of fibromyalgia syndrome. Future Rheumatol. 2008, 3, 475–483.
- 95. Tracy, L.; Ioannou, L.; Baker, K.; Gibson, S.; Georgiou-Karistianis, N.; Giummarra, M. Meta-analytic evidence for decreased heart rate variability in chronic pain implicating parasympathetic nervous system dysregulation. *Pain* **2015**, *157*, 7–29.
- 96. Hallman, D.; Lyskov, E. Autonomic regulation, physical activity and perceived stress in subjects with musculoskeletal pain: 24-hour ambulatory monitoring. *Int. J. Psychophysiol.* **2012**, *86*, 276–282.
- 97. Tracy, L.; Gibson, S.; Labuschagne, I.; Georgiou-Karistianis, N.; Giummarra, M. Intranasal oxytocin reduces heart rate variability during a mental arithmetic task: A randomised, double-blind, placebo-controlled cross-over study. *Prog. Neuro-Psychopharmacol. Biol. Psychiatry* 2017, *81*, 408–415.
- Clapp, J.; Croarkin, S.; Dolphin, C.; Lyons, S. Heart rate variability: A biomarker of dairy calf welfare. *Anim. Prod. Sci.* 2015, 55, 1289–1294.
- 99. Goto, Y. HRV-spectral analysis of pain, by 3D evaluation and by balance index in the pain rehabilitation field. *Jpn. J. Anesthesiol.* **2015**, *64*, 741–751.
- 100. Goto, Y.; Komatsu, T.; Nakagawa, T.; Noguchi, H.; Tsuda, T. A new index of the autonomic nervous balance and its clinical usefulness; A power law 1/f like scale of a spectral analysis of the heart rate variability. *J. Clin. Monit. Comput.* 2007, 21, 21–30.
- 101. Morikawa, Y.; Takamoto, K.; Nishimaru, H.; Taguchi, T.; Urakawa, S.; Sakai, S.; Ono, T.; Nishijo, H. Compression at myofascial trigger point on chronic neck pain provides pain relief through the prefrontal cortex and autonomic nervous system: A pilot study. *Front. Neurosci.* 2017, *11*, 186.
- 102. Shiro, Y.; Arai, Y.C.; Matsubara, T.; Isogai, S.; Ushida, T. Effect of muscle load tasks with maximal isometric contractions on oxygenation of the trapezius muscle and sympathetic nervous activity in females with chronic neck and shoulder pain. *BMC Musculoskelet. Disord.* **2012**, *13*, 146.
- 103. Södervall, J.; Karppinen, J.; Puolitaival, J.; Kyllönen, E.; Kiviniemi, A.; Tulppo, M.; Hautala, A. Heart rate variability in sciatica patients referred to spine surgery: A case control study. *BMC Musculoskelet. Disord.* 2013, 14, 149.
- 104. Gevirtz, R. Autonomic Nervous System Markers for Psychophysiological, Anxiety, and Physical Disorders; Oxford University Press: Oxford, UK, 2011.
- 105. Walker, L.; Stone, A.; Smith, C.; Bruehl, S.; Garber, J.; Puzanovova, M.; Diedrich, A. Interacting influences of gender and chronic pain status on parasympathetically mediated heart rate variability in adolescents and young adults. *Pain* 2017, 158, 1509–1516.
- 106. Burr, R.; Heitkemper, M.; Jarrett, M.; Cain, K. Comparison of autonomic nervous system indices based on abdominal pain reports in women with irritable bowel syndrome. *Biol. Res. Nurs.* 2000, *2*, 97–106.
- 107. Cain, K.; Jarrett, M.; Burr, R.; Hertig, V.; Heitkemper, M. Heart rate variability is related to pain severity and predominant bowel pattern in women with irritable bowel syndrome. *Neurogastroenterol. Motil.* **2007**, *19*, 110–118.
- Davydov, D.; Naliboff, B.; Shahabi, L.; Shapiro, D. Baroreflex mechanisms in Irritable Bowel Syndrome: Part I. Traditional indices. *Physiol. Behav.* 2016, 157, 102–108.
- 109. Momota, Y.; Takano, H.; Kani, K.; Matsumoto, F.; Motegi, K.; Aota, K.; Yamamura, Y.; Omori, M.; Tomioka, S.; Azuma, M. Frequency analysis of heart rate variability: A useful assessment tool of linearly polarized near-infrared irradiation to stellate ganglion area for burning mouth syndrome. *Pain Med.* 2013, 14, 351–357.
- 110. Quartana, P.; Campbell, C.; Edwards, R. Pain catastrophizing a critical review. *Expert Rev. Neurother.* **2009**, *9*, 745–758.
- 111. Koenig, J.; De Kooning, M.; Bernardi, A.; Williams, D.; Nijs, J.; Thayer, J.; Daenen, L. Lower Resting State Heart Rate Variability Relates to High Pain Catastrophizing in Patients with Chronic Whiplash-Associated Disorders and Healthy Controls. *Pain Pract.* 2016, 16, 1048–1053.
- 112. Edelhäuser, F.; Minnerop, A.; Trapp, B.; Büssing, A.; Cysarz, D. Eurythmy therapy increases specific oscillations of heart rate variability. *BMC Complement. Altern. Med.* **2015**, *15*, 167.
- 113. Giggins, O.; Persson, U.; Caulfield, B. Biofeedback in rehabilitation. J. NeuroEng. Rehabil. 2013, 10, 60.

- Weeks, D.; Whitney, A.; Tindall, A.; Carter, G. Pilot randomized trial comparing intersession scheduling of biofeedback results to individuals with chronic pain: Influence on psychologic function and pain intensity. *Am. J. Phys. Med. Rehabil.* 2015, *94*, 869–878.
- 115. Kampusch, S.; Kaniusas, E.; Széles, J. New approaches in multi-punctual percutaneous stimulation of the auricular vagus nerve. In Proceedings of the 2013 6th International IEEE/EMBS Conference on Neural Engineering (NER), San Diego, CA, USA, 6–8 November 2013; pp. 263–266.
- 116. Kampusch, S.; Kaniusas, E.; Széles, J. Modulation of Muscle Tone and Sympathovagal Balance in Cervical Dystonia Using Percutaneous Stimulation of the Auricular Vagus Nerve. *Artif. Organs* **2015**, *39*, E202–E212.
- 117. Shafiq, H.; McGregor, C.; Murphy, B. The Impact of Cervical Manipulation on Heart Rate Variability. In Proceedings of the 2014 36th Annual International Conference of the IEEE Engineering in Medicine and Biology Society, EMBC 2014, Chicago, IL, USA, 26–30 August 2014; pp. 3406–3409.
- 118. Caza-Szoka, M.; Massicotte, D.; Nougarou, F. Naive Bayesian learning for small training samples: Application on chronic Low Back Pain diagnostic with sEMG sensors. In Proceedings of the Conference Record - IEEE Instrumentation and Measurement Technology Conference, Pisa, Italy, 11–14 May 2015; pp. 470–475.
- 119. Telles, S.; Sharma, S.; Gupta, R.; Bhardwaj, A.; Balkrishna, A. Heart rate variability in chronic low back pain patients randomized to yoga or standard care. *BMC Complement. Altern. Med.* **2016**, *16*, 279.
- 120. Garland, E.; Manusov, E.; Froeliger, B.; Kelly, A.; Williams, J.; Howard, M. Mindfulness-oriented recovery enhancement for chronic pain and prescription opioid misuse: Results from an early-stage randomized controlled trial. *J. Consult. Clin. Psychol.* **2014**, *82*, 448–459.
- 121. Garland, E.; Froeliger, B.; Howard, M. Effects of Mindfulness-Oriented Recovery Enhancement on reward responsiveness and opioid cue-reactivity. *Psychopharmacology* **2014**, *231*, 3229–3238.
- 122. Garland, E.; Froeliger, B.; Howard, M. Allostatic dysregulation of natural reward processing in prescription opioid misuse: Autonomic and attentional evidence. *Biol. Psychol.* **2015**, *105*, *124–129*.
- 123. Perruchoud, C.; Buchser, E.; Johanek, L.; Aminian, K.; Paraschiv-Ionescu, A.; Taylor, R. Assessment of physical activity of patients with chronic pain. *Neuromodulation* **2014**, *17*, 42–47.
- 124. Wang, C.; Olugbade, T.; Mathur, A.; De Williams, A.; Lane, N.; Bianchi-Berthouze, N. Recurrent Network based Automatic Detection of Chronic Pain Protective Behavior using MoCap and sEMG data. In Proceedings of the Proceedings International Symposium on Wearable Computers (ISWC), London, UK, 9–13 September 2019; pp. 225–230.
- 125. Nordstoga, A.; Meisingset, I.; Vasseljen, O.; Nilsen, T.; Unsgaard-Tondel, M. Longitudinal associations of kinematics and fear-avoidance beliefs with disability, work ability and pain intensity in persons with low back pain. *Musculoskelet. Sci. Pract.* **2019**, *41*, 49–54.
- Dekker-Van Weering, M.; Vollenbroek-Hutten, M.; Hermens, H. A pilot study-the potential value of an activity-based feedback system for treatment of individuals with chronic lower back pain. *Disabil. Rehabil.* 2015, *37*, 2250–2256.
- 127. Boston, J.; Rudy, T.; Mercer, S.; Kubinski, J. A measure of body movement coordination during repetitive dynamic lifting. *IEEE Trans. Rehabil. Eng.* **1993**, *1*, 137–144.
- 128. Slaboda, J.; Boston, J.; Rudy, T.; Lieber, S.; Rasetshwane, D. The use of splines to calculate jerk for a lifting task involving chronic lower back pain patients. *IEEE Trans. Neural Syst. Rehabil. Eng.* **2005**, *13*, 406–414.
- 129. Treleaven, J.; Takasaki, H.; Grip, H. Altered trunk head co-ordination in those with persistent neck pain. *Musculoskelet. Sci. Pract.* **2019**, *39*, 45–50.
- 130. Plaas, H.; Sudhaus, S.; Willburger, R.; Hasenbring, M. Physical activity and low back pain: The role of subgroups based on the avoidance-endurance model. *Disabil. Rehabil.* **2014**, *36*, 749–755.
- Schaller, A.; Rudolf, K.; Dejonghe, L.; Grieben, C.; Froboese, I. Influencing Factors on the Overestimation of Self-Reported Physical Activity: A Cross-Sectional Analysis of Low Back Pain Patients and Healthy Controls. *BioMed Res. Int.* 2016, 2016, 497213.
- 132. Bavan, L.; Surmacz, K.; Beard, D.; Mellon, S.; Rees, J. Adherence monitoring of rehabilitation exercise with inertial sensors: A clinical validation study. *Gait Posture* **2019**, *70*, 211–217.
- 133. Carvalho, F.; Morelhão, P.; Franco, M.; Maher, C.; Smeets, R.; Oliveira, C.; Freitas Júnior, I.; Pinto, R. Reliability and validity of two multidimensional self-reported physical activity questionnaires in people with chronic low back pain. *Musculoskelet. Sci. Pract.* **2017**, *27*, 65–70.

- 134. Van Weering, M.; Vollenbroek-Hutten, M.; Hermens, H. The relationship between objectively and subjectively measured activity levels in people with chronic low back pain. *Clin. Rehabil.* **2011**, *25*, 256–263.
- 135. Huijnen, I.; Verbunt, J.; Peters, M.; Delespaul, P.; Kindermans, H.; Roelofs, J.; Goossens, M.; Seelen, H. Do depression and pain intensity interfere with physical activity in daily life in patients with Chronic Low Back Pain? *Pain* **2010**, *150*, 161–166.
- 136. Schaller, A.; Rudolf, K.; Arndt, F.; Froboese, I. Assessing Physical Activity in Low Back Pain Patients after Inpatient Rehabilitation: A Comparison of a Direct Versus a Self-report Measure. *Phys. Med. Rehabilitationsmed. Kurortmed.* 2016, 26, 71–78.
- 137. Khan, Z.; Sohn, W. Abnormal human activity recognition system based on R-transform and independent component features for elderly healthcare. *J. Chin. Inst. Eng. Trans. Chin. Inst. Eng.* **2013**, *36*, 441–451.
- 138. Khan, Z.; Sohn, W. A hierarchical abnormal human activity recognition system based on R-transform and kernel discriminant analysis for elderly health care. *Computing* **2013**, *95*, 109–127.
- Raijmakers, B.; Nieuwenhuizen, M.; Beckerman, H.; de Groot, S. Differences in the course of daily activity level between persons with and without chronic pain. *Am. J. Phys. Med. Rehabil. Assoc. Acad. Phys.* 2015, 94, 101–109; quiz 110–113.
- 140. Yang, C.C.; Hsu, Y.L. A review of accelerometry-based wearable motion detectors for physical activity monitoring. *Sensors* **2010**, *10*, 7772–7788.
- 141. Leininger, B.; Schulz, C.; Gao, Z.; Bronfort, G.; Evans, R.; Pope, Z.; Zeng, N.; Haas, M. Accelerometer-determined physical activity and clinical low back pain measures in adolescents with chronic or subacute recurrent low back pain. *J. Orthop. Sports Phys. Ther.* **2017**, *47*, 769–774.
- 142. Murphy, S.; Schepens Niemiec, S.; Lyden, A.; Kratz, A. Pain, Fatigue, and Physical Activity in Osteoarthritis: The Moderating Effects of Pain- and Fatigue-Related Activity Interference. *Arch. Phys. Med. Rehabil.* 2016, 97, S201–S209.
- 143. Sorensen, C.; Hastings, M.; Lang, C.; McGill, J.; Clark, B.; Bohnert, K.; Mueller, M. Relationship of shoulder activity and skin intrinsic fluorescence with low level shoulder pain and disability in people with type 2 diabetes. *J. Diabetes Complic.* **2017**, *31*, 983–987.
- 144. Atallah, L.; Aziz, O.; Gray, E.; Lo, B.; Yang, G.Z. An ear-worn sensor for the detection of gait impairment after abdominal surgery. *Surg. Innov.* **2013**, *20*, 86–94.
- 145. Norgaard, M.; Lomholt, J.; Thastum, M.; Herlin, M.; Twilt, M.; Herlin, T. Accelerometer-assessed daily physical activity in relation to pain cognition in juvenile idiopathic arthritis. *Scand. J. Rheumatol.* **2017**, 46, 22–26.
- 146. Jørgensen, M.; Korshøj, M.; Lagersted-Olsen, J.; Villumsen, M.; Mortensen, O.; Skotte, J.; Søgaard, K.; Madeleine, P.; Thomsen, B.; Holtermann, A. Physical activities at work and risk of musculoskeletal pain and its consequences: Protocol for a study with objective field measures among blue-collar workers. *BMC Musculoskelet. Disord.* 2013, 14, 213.
- 147. Reneman, M.; De Vries, H.; Van Den Hengel, E.; Brouwer, S.; Van Der Woude, L. Different level, but a similar day pattern of physical activity in workers and sick-listed people with chronic nonspecific musculoskeletal pain. *Arch. Phys. Med. Rehabil.* **2012**, *93*, 1864–1867.
- 148. Barkallah, E.; Otis, M.; Ngomo, S.; Heraud, M. Measuring operator's pain: Toward evaluating musculoskeletal disorder at work. In Proceedings of the 2015 IEEE International Conference on Systems, Man, and Cybernetics (SMC 2015), Kowloon, Hong Kong, 9–12 October 2015; pp. 2366–2371.
- 149. Pinheiro, M.; Ferreira, M.; Refshauge, K.; Hopper, J.; Maher, C.; Hartvigsen, J.; Koes, B.; Hübscher, M.; Ferreira, P. Protective and harmful effects of physical activity for low back pain: A protocol for the AUstralian Twin BACK Pain (AUTBACK) feasibility study. *Twin Res. Hum. Genet.* 2016, 19, 502–509.
- Hallman, D.; Gupta, N.; Mathiassen, S.; Holtermann, A. Association between objectively measured sitting time and neck-shoulder pain among blue-collar workers. *Int. Arch. Occup. Environ. Health* 2015, *88*, 1031–1042.
- 151. Munch Nielsen, C.; Gupta, N.; Knudsen, L.; Holtermann, A. Association of objectively measured occupational walking and standing still with low back pain: a cross-sectional study. *Ergonomics* **2017**, *60*, 118–126.
- 152. Lunde, L.K.; Koch, M.; Knardahl, S.; Veiersted, K. Associations of objectively measured sitting and standing with low-back pain intensity: A 6-month follow-up of construction and healthcare workers. *Scand. J. Work Environ. Health* **2017**, *43*, 269–278.

- 153. Balaguier, R.; Madeleine, P.; Rose-Dulcina, K.; Vuillerme, N. Trunk kinematics and low back pain during pruning among vineyard workers—A field study at the Chateau Larose-Trintaudon. *PLoS ONE* **2017**, *12*, e0175126.
- 154. Kaleth, A.; Slaven, J.; Ang, D. Does increasing steps per day predict improvement in physical function and pain interference in adults with fibromyalgia? *Arthritis Care Res.* **2014**, *66*, 1887–1894.
- 155. Hallman, D.; Birk Jørgensen, M.; Holtermann, A. Objectively measured physical activity and 12-month trajectories of neck-shoulder pain in workers: A prospective study in DPHACTO. *Scand. J. Public Health* **2017**, *45*, 288–298.
- 156. Motl, R.; McAuley, E.; Snook, E.; Gliottoni, R. Physical activity and quality of life in multiple sclerosis: Intermediary roles of disability, fatigue, mood, pain, self-efficacy and social support. *Psychol. Health Med.* 2009, 14, 111–124.
- 157. Matsubara, T.; Arai, Y.C.; Kazuhiro, S.; Sato, J.; Osuga, T.; Nishihara, M.; Ushida, T. Effects of cognitive-behavioral therapy on pain intensity and level of physical activity in Japanese patients with chronic pain- a preliminary quasiexperimental study. *J. Phys. Ther.* **2010**, *1*, 49–57.
- 158. Andrews, N.; Strong, J.; Meredith, P.; D'Arrigo, R. Association between physical activity and sleep in adults with chronic pain: A momentary, within-person perspective. *Phys. Ther.* **2014**, *94*, 499–510.
- 159. Van Hees, V.; Pias, M.; Taherian, S.; Ekelund, U.; Brage, S. A method to compare new and traditional accelerometry data in physical activity monitoring. In Proceedings of the IEEE International Symposium on "A World of Wireless, Mobile and Multimedia Networks", Montreal, QC, Canada, 14–17 June 2010.
- 160. Lord, S.; Chastin, S.; McInnes, L.; Little, L.; Briggs, P.; Rochester, L. Exploring patterns of daily physical and sedentary behaviour in community-dwelling older adults. *Age Ageing* **2011**, *40*, 205–210.
- Gerdhem, P.; Dencker, M.; Ringsberg, K.; Akesson, K. Accelerometer-measured daily physical activity among octogenerians: Results and associations to other indices of physical performance and bone density. *Eur. J. Appl. Physiol.* 2008, *102*, 173–180.
- 162. Dinh, A.; Shi, Y.; Teng, D.; Ralhan, A.; Chen, L.; Dal Bello-Haas, V.; Basran, J.; Ko, S.B.; McCrowsky, C. A fall and near-fall assessment and evaluation system. *Open Biomed. Eng. J.* **2009**, *3*, 1–7.
- 163. Makikawa, M.; Asajima, S.; Shibuya, K.; Tokue, R.; Shinohara, H. Portable physical activity monitoring system for the evaluation of activity of the aged in daily life. In Proceedings of the Annual International Conference of the IEEE Engineering in Medicine and Biology, Houston, TX, USA, 23–26 October 2002; Volume 3, pp. 1908–1909.
- 164. Plasqui, G.; Westerterp, K. Physical activity assessment with accelerometers: An evaluation against doubly labeled water. *Obesity* **2007**, *15*, 2371–2379.
- 165. Copeland, J.; Esliger, D. Accelerometer assessment of physical activity in active, healthy older adults. *J. Aging Phys. Act.* **2009**, *17*, 17–30.
- 166. Naranjo-Hernández, D.; Roa, L.; Reina-Tosina, J.; Estudillo-Valderrama, M. SoM: A smart sensor for human activity monitoring and assisted healthy ageing. *IEEE Trans. Biomed. Eng.* **2012**, *59*, 3177–3184.
- 167. Najafi, B.; Aminian, K.; Paraschiv-Ionescu, A.; Loew, F.; Bula, C.; Robert, P. Ambulatory system for human motion analysis using a kinematic sensor: Monitoring of daily physical activity in the elderly. *IEEE Trans. Biomed. Eng.* 2003, 50, 711–723.
- Yang, C.C.; Hsu, Y.L. Algorithm design for real-time physical activity identification with accelerometry measurement. In Proceedings of the Industrial Electronics Conference, Taipei, Taiwan, 5–8 November 2007; pp. 2996–3000.
- 169. Tadayon, P.; Felderhoff, T.; Staude, G.; Knopp, A. Mobile System for the Prevention, Diagnosis, and Personalized Treatment of Neck Pain under a Patient's Everyday Life Circumstances. *Curr. Direct. Biomed. Eng.* 2019, *5*, 257–260.
- 170. M. Guralnik, M.; Simonsick, E.; Ferrucci, L.; J. Glynn, R.; F. Berkman, L.; Blazer, D.; A. Scherr, P.; B. Wallace, R. A Short Physical Performance Battery Assessing Lower Extremity Function: Association With Self-Reported Disability and Prediction of Mortality and Nursing Home Admission. J. Gerontol. 1994, 49, M85–94.
- 171. Myers, D.; Weiss, A.; Rollins, M.; Lam, W. Towards remote assessment and screening of acute abdominal pain using only a smartphone with native accelerometers. *Sci. Rep.* **2017**, *7*, 12750..
- 172. Cushing, C.; Kichline, T.; Blossom, J.; Friesen, C.; Schurman, J. Tailoring individualized evaluation of pediatric abdominal pain using ecological momentary assessment (EMA): a pilot study testing feasibility and acceptability. *Clin. J. Pain* **2019**, *35*, 859–868.

- 173. Madeleine, P.; Søgaard, K.; Holtermann, A.; Samani, A. Level of self-reported neck/shoulder pain and biomechanical workload in cleaners. *Work* **2012**, *41*, 447–452.
- 174. Sliepen, M.; Mauricio, E.; Lipperts, M.; Grimm, B.; Rosenbaum, D. Objective assessment of physical activity and sedentary behaviour in knee osteoarthritis patients - Beyond daily steps and total sedentary time. BMC Musculoskelet. Disord. 2018, 19, 64.
- 175. Huijnen, I.; Verbunt, J.; Peters, M.; Smeets, R.; Kindermans, H.; Roelofs, J.; Goossens, M.; Seelen, H. Differences in activity-related behaviour among patients with chronic low back pain. *Eur. J. Pain* **2011**, *15*, 748–755.
- 176. van Weering, M.; Vollenbroek-Hutten, M.; Tönis, T.; Hermens, H. Daily physical activities in chronic lower back pain patients assessed with accelerometry. *Eur. J. Pain* **2009**, *13*, 649–654.
- 177. Naugle, K.; Ohlman, T.; Naugle, K.; Riley, Z.; Keith, N. Physical activity behavior predicts endogenous pain modulation in older adults. *Pain* **2017**, *158*, 383–390.
- 178. Robbins, S.; Birmingham, T.; Callaghan, J.; Jones, G.; Chesworth, B.; Maly, M. Association of pain with frequency and magnitude of knee loading in knee osteoarthritis. *Arthritis Care Res.* **2011**, *63*, 991–997.
- 179. Liszka-Hackzell, J.; Martin, D. An analysis of the relationship between activity and pain in chronic and acute low back pain. *Anesth. Analg.* **2004**, *99*, 477–481.
- 180. Kataoka, H.; Ikemoto, T.; Yoshimura, A.; Shibuya, M.; Goto, K.; Yamashita, J.; Morita, K.; Sakamoto, J.; Nakano, J.; Okita, M. Association of early physical activity time with pain, activities of daily living, and progression of vertebral body collapse in patients with vertebral compression fractures. *Eur. J. Phys. Rehabil. Med.* 2017, 53, 366–376.
- 181. Van Koppen, L.; Zandwijk, P.; Van Mameren, H.; Mesters, I.; Winkens, B.; De Bie, R. Patients adherence to a walking programme for non-specific low back pain. *Eur. J. Physiother.* **2016**, *18*, 103–115.
- 182. Sani, S.; Massie, S.; Wiratunga, N.; Cooper, K. Learning deep and shallow features for human activity recognition. *Lect. Notes Comput. Sci.* 2017, 10412 LNAI, 469–482.
- 183. Sani, S.; Wiratunga, N.; Massie, S.; Cooper, K. kNN sampling for personalised human activity recognition. *Lect. Notes Comput. Sci.* **2017**, *10339 LNAI*, 330–344.
- 184. Li, H.; Peng, F.; Du, W.; Wang, L. Quantitative Evaluation for Different Level of Pain in Patients with Low Back Pain compared with Healthy Control. In Proceedings of the International Conference on Signal Processing (ICSP), Beijing, China, 12–16 August 2018; pp. 944–949.
- 185. Ledowski, T.; Bromilow, J.; Paech, M.; Storm, H.; Hacking, R.; Schug, S. Monitoring of skin conductance to assess postoperative pain intensity. *Br. J. Anaesth.* **2006**, *97*, 862–865.
- 186. Roa, L.; Naranjo, D.; Reina-Tosina, J.; others. Applications of bioimpedance to end stage renal disease (ESRD). *Stud. Comput. Intell.* **2013**, 404, 689–769.
- Ledowski, T.; Bromilow, J.; Wu, J.; Paech, M.; Storm, H.; Schug, S. The assessment of postoperative pain by monitoring skin conductance: Results of a prospective study. *Anaesthesia* 2007, 62, 989–993.
- Bartley, E.; Rhudy, J.; Williams, A. Experimental assessment of affective processing in fibromyalgia. *J. Pain* 2009, 10, 1151–1160.
- Lush, E.; Salmon, P.; Floyd, A.; Studts, J.; Weissbecker, I.; Sephton, S. Mindfulness meditation for symptom reduction in Fibromyalgia: Psychophysiological correlates. J. Clin. Psychol. Med. Settings 2009, 16, 200–207.
- 190. Thieme, K.; Turk, D. Heterogeneity of psychophysiological stress responses in fibromyalgia syndrome patients. *Arthritis Res. Ther.* **2006**, *8*, R9.
- 191. Thieme, K.; Turk, D.; Gracely, R.; Maixner, W.; Flor, H. The relationship among psychological and psychophysiological characteristics of fibromyalgia patients. *J. Pain* **2015**, *16*, 186–196.
- 192. Thieme, K.; Rose, U.; Pinkpank, T.; Spies, C.; Turk, D.; Flor, H. Psychophysiological responses in patients with fibromyalgia syndrome. *J. Psychosom. Res.* **2006**, *61*, 671–679.
- 193. Glombiewski, J.; Tersek, J.; Rief, W. Muscular reactivity and specificity in chronic back pain patients. *Psychosom. Med.* **2008**, *70*, 125–131.
- 194. Schneider, C.; Palomba, D.; Flor, H. Pavlovian conditioning of muscular responses in chronic pain patients: Central and peripheral correlates. *Pain* **2004**, *112*, 239–247.
- 195. Flor, H.; Diers, M.; Birbaumer, N. Peripheral and electrocortical responses to painful and non-painful stimulation in chronic pain patients, tension headache patients and healthy controls. *Neurosci. Lett.* **2004**, *361*, 147–150.

- 196. Spence, S.; Sharpe, L.; Newton-John, T.; Champion, D. An investigation of symptom-specific muscle hyperreactivity in upper extremity cumulative trauma disorder. *Clin. J. Pain* **2001**, *17*, 119–128.
- 197. Knost, B.; Flor, H.; Birbaumer, N.; Schugens, M. Learned maintenance of pain: Muscle tension reduces central nervous system processing of painful stimulation in chronic and subchronic pain patients. *Psychophysiology* 1999, 36, 755–764.
- 198. McCracken, L.; Goetsch, V.; Semenchuk, E. Coping with pain produced by physical activity in persons with chronic low back pain: Immediate assessment following a specific pain event. *Behav. Med.* **1998**, *24*, 29–34.
- 199. Yamamoto, N.; Itoi, E.; Minagawa, H.; Seki, N.; Abe, H.; Shimada, Y.; Okada, K. Objective evaluation of shoulder pain by measuring skin impedance. *Orthopedics* **2006**, *29*, 1121–1123.
- 200. Glombiewski, J.; Riecke, J.; Holzapfel, S.; Rief, W.; König, S.; Lachnit, H.; Seifart, U. Do patients with chronic pain show autonomic arousal when confronted with feared movements? An experimental investigation of the fear-avoidance model. *Pain* **2015**, *156*, 547–554.
- Jenewein, J.; Erni, J.; Moergeli, H.; Grillon, C.; Schumacher, S.; Mueller-Pfeiffer, C.; Hassanpour, K.; Seiler, A.; Wittmann, L.; Schnyder, U.; et al. Altered pain perception and fear-learning deficits in subjects with posttraumatic stress disorder. *J. Pain* 2016, *17*, 1325–1333.
- 202. Terry, E.; Delventura, J.; Bartley, E.; Vincent, A.; Rhudy, J. Emotional modulation of pain and spinal nociception in persons with major depressive disorder (MDD). *Pain* **2013**, *154*, 2759–2768.
- 203. Williams, A.; Rhudy, J. Emotional modulation of autonomic responses to painful trigeminal stimulation. *Int. J. Psychophysiol.* **2009**, *71*, 242–247.
- 204. Tomé-Pires, C.; Miro, J. Electrodermal responses and memory recall in migraineurs and headache-free controls. *Eur. J. Pain* **2014**, *18*, 1298–1306.
- 205. Icenhour, A.; Langhorst, J.; Benson, S.; Schlamann, M.; Hampel, S.; Engler, H.; Forsting, M.; Elsenbruch, S. Neural circuitry of abdominal pain-related fear learning and reinstatement in irritable bowel syndrome. *Neurogastroenterol. Motil.* 2015, 27, 114–127.
- 206. O'Leary, H.; Marschik, P.; Khwaja, O.; Ho, E.; Barnes, K.; Clarkson, T.; Bruck, N.; Kaufmann, W. Detecting autonomic response to pain in Rett syndrome. *Dev. Neurorehabil.* **2017**, *20*, 108–114.
- 207. Bonnet, A.; Naveteur, J. Electrodermal responses to words in chronic low back pain patients: A comparison between pain descriptors, other emotional words, and neutral words. *Clin. J. Pain* **2006**, *22*, 686–691.
- 208. Flor, H.; Knost, B.; Birbaumer, N. Processing of pain- and body-related verbal material in chronic pain patients: Central and peripheral correlates. *Pain* **1997**, *73*, 413–421.
- 209. Ferrari, R.; Fipaldini, E.; Leone, C. Psychophysiological reactivity to imagined stress in patients with chronic pain. *Psicoter. Cogn. Comport.* 2004, *10*, 179–196.
- 210. Block, A. An investigation of the response of the spous to chronic pain behavior. *Psychosom. Med.* **1981**, 43, 415–422.
- 211. Apkarian, A.; Sosa, Y.; Krauss, B.; Thomas, P.; Fredrickson, B.; Levy, R.; Harden, R.; Chialvo, D. Chronic pain patients are impaired on an emotional decision-making task. *Pain* **2004**, *108*, 129–136.
- Elvemo, N.A.; Nilsen, K.; Landro, N.; Borchgrevink, P.; Haberg, A. Patients with chronic pain lack somatic markers during decision-making. *J. Pain Res.* 2014, 7, 425–437.
- 213. Chalaye, P.; Goffaux, P.; Lafrenaye, S.; Marchand, S. Respiratory effects on experimental heat pain and cardiac activity. *Pain Med.* **2009**, *10*, 1334–1340.
- Busch, V.; Magerl, W.; Kern, U.; Haas, J.; Hajak, G.; Eichhammer, P. The effect of deep and slow breathing on pain perception, autonomic activity, and mood processing-an experimental study. *Pain Med.* 2012, 13, 215–228.
- 215. Larsen, L.; Hirata, R.; Graven-Nielsen, T. Experimental Low Back Pain Decreased Trunk Muscle Activity in Currently Asymptomatic Recurrent Low Back Pain Patients During Step Tasks. *J. Pain* **2018**, *19*, 542–551.
- 216. Sai, C.; Mokhtar, N.; Yip, H.; Bak, L.; Hasan, M.; Arof, H.; Cumming, P.; Mat Adenan, N. Objective identification of pain due to uterine contraction during the first stage of labour using continuous EEG signals and SVM. *Sad. Acad. Proc. Eng. Sci.* **2019**, *44*, 87.
- 217. Gram, M.; Graversen, C.; Nielsen, A.; Arendt-Nielsen, T.; Mørch, C.D.; Andresen, T.; Drewes, A.M. A novel approach to pharmaco-EEG for investigating analgesics: Assessment of spectral indices in single-sweep evoked brain potentials. *Br. J. Clin. Pharmacol.* **2013**, *76*, 951–963.
- 218. Giehl, J.; Meyer-Brandis, G.; Kunz, M.; Lautenbacher, S. Responses to tonic heat pain in the ongoing EEG under conditions of controlled attention. *Somatosens. Mot. Res.* **2014**, *31*, 40–48.

- 219. Nir, R.R.; Sinai, A.; Raz, E.; Sprecher, E.; Yarnitsky, D. Pain assessment by continuous EEG: Association between subjective perception of tonic pain and peak frequency of alpha oscillations during stimulation and at rest. *Brain Res.* **2010**, *1344*, 77–86.
- 220. Haag, A.; Goronzy, S.; Schaich, P.; Williams, J., Emotion Recognition Using Bio-sensors: First Steps towards an Automatic System. In Affective Dialogue Systems: Tutorial and Research Workshop, ADS 2004, Kloster Irsee, Germany, 14–16 June 2004; André, E.; Dybkjær, L.; Minker, W.; Heisterkamp, P., Eds.; Springer: Berlin/ Heidelberg, Germany, 2004; pp. 36–48.
- 221. Becker, S.; Bergamo, F.; Schnake, K.; Schreyer, S.; Rembitzki, I.; Disselhorst-Klug, C. The relationship between functionality and erector spinae activity in patients with specific low back pain during dynamic and static movements. *Gait Posture* **2018**, *66*, 208–213.
- 222. Choi, H.S. EMG sensor system for neck fatigue assessment using RF wireless power transmission. In Proceedings of the 3rd International Conference on Computational Intelligence and Applications (ICCIA 2018), Hong Kong, China, 28–30 July 2018; pp. 219–222.
- 223. Bonato, P.; Mork, P.; Sherrill, D.; Westgaard, R. Data mining of motor patterns recorded with wearable technology. *IEEE Eng. Med. Biol. Mag.* **2003**, *22*, 110–119.
- 224. Watson, P.; Booker, C.; Main, C.; Chen, A. Surface electromyography in the identification of chronic low back pain patients: The development of the flexion relaxation ratio. *Clin. Biomech.* **1997**, *12*, 165–171.
- 225. Iwasaki, L.; Gonzalez, Y.; Liu, H.; Marx, D.; Gallo, L.; Nickel, J. A pilot study of ambulatory masticatory muscle activities in temporomandibular joint disorders diagnostic groups. *Orthod. Craniofac. Res.* 2015, 18, 146–155.
- 226. Ahern, D.; Follick, M.; Council, J.; Laser-Wolston, N.; Litchman, H. Comparison of lumbar paravertebral EMG patterns in chronic low back pain patients and non-patient controls. *Pain* **1988**, *34*, 153–160.
- 227. Caza-Szoka, M.; Massicotte, D.; Nougarou, F.; Descarreaux, M. Surrogate analysis of fractal dimensions from SEMG sensor array as a predictor of chronic low back pain. In Proceedings of the Annual International Conference of the IEEE Engineering in Medicine and Biology Society, EMBS, Orlando, FL, USA, 16–20 August 2016; pp. 6409–6412.
- 228. Biurrun Manresa, J.; Nguyen, G.; Curatolo, M.; Moeslund, T.; Andersen, O. Probabilistic model for individual assessment of central hyperexcitability using the nociceptive withdrawal reflex: A biomarker for chronic low back and neck pain. *BMC Neurosci.* 2013, *14*, 110.
- 229. Nguyen, G.; Manresa, J.; Curatolo, M.; Moeslund, T.; Andersen, O. A prediction model for differentiating chronic pain patients and healthy subjects based on withdrawal reflex EMG signals. In Proceedings of the 2011 5th International IEEE/EMBS Conference on Neural Engineering (NER 2011), Cancun, Mexico, 27 April–1 May 2011; pp. 92–95.
- 230. Chang Chien, J.R.; Lin, G.H.; Hsu, A.T. Implementation of a portable electronic system for providing pain relief to patellofemoral pain syndrome patients. *Rev. Sci. Instrum.* **2011**, *82*, 1171–1174.
- 231. Yang, G.; Jiang, M.; Ouyang, W.; Ji, G.; Xie, H.; Rahmani, A.; Liljeberg, P.; Tenhunen, H. IoT-Based Remote Pain Monitoring System: From Device to Cloud Platform. *IEEE J. Biomed. Health Inform.* **2018**, *22*, 1711–1719.
- 232. Schulz, E.; Zherdin, A.; Tiemann, L.; Plant, C.; Ploner, M. Decoding an individual's sensitivity to pain from the multivariate analysis of EEG data. *Cereb. Cortex* **2012**, *22*, 1118–1123.
- 233. Nir, R.R.; Sinai, A.; Moont, R.; Harari, E.; Yarnitsky, D. Tonic pain and continuous EEG: Prediction of subjective pain perception by alpha-1 power during stimulation and at rest. *Clin. Neurophysiol.* 2012, 123, 605–612.
- 234. Dowman, R.; Rissacher, D.; Schuckers, S. EEG indices of tonic pain-related activity in the somatosensory cortices. *Clin. Neurophysiol.* **2008**, *119*, 1201–1212.
- 235. Keight, R.; Al-Jumeily, D.; Hussain, A.J.; Al-Jumeily, M.; Mallucci, C. Towards the discrimination of primary and secondary headache: An intelligent systems approach. In Proceedings of the 2017 International Joint Conference on Neural Networks (IJCNN), Anchorage, AK, USA, 14–19 May 2017; pp. 2768–2775.
- 236. Karydis, T.; Foster, S.; Mershin, A. Self-Calibrating Protocols as diagnostic AIDS for personal medicine, neurological conditions and pain assessment. In Proceedings of the ACM International Conference Proceeding Series, Corfu, Greece, 29 June–1 July 2016.
- 237. Hadjistavropoulos, T.; Craig, K.; Duck, S.; others. A Biopsychosocial Formulation of Pain Communication. *Psychol. Bull.* **2011**, *137*, 910–939.

- 238. Wu, L.; Kirmse, U.; Flaisch, T.; Boiandina, G.; Kenter, A.; Schupp, H. Empathy, pain and attention: Cues that predict pain stimulation to the partner and the self capture visual attention. *Front. Hum. Neurosci.* **2017**, *11*, 465.
- 239. Bargshady, G.; Soar, J.; Zhou, X.; Deo, R.; Whittaker, F.; Wang, H. A joint deep neural network model for pain recognition from face. In Proceedings of the 2019 IEEE 4th International Conference on Computer and Communication Systems (ICCCS 2019), Singapore, 23–25 February 2019; pp. 52–56.
- 240. Haque, M.; Bautista, R.; Noroozi, F.; Kulkarni, K.; Laursen, C.B.; Irani, R.; Bellantonio, M.; Escalera, S.; Anbarjafari, G.; Nasrollahi, K.; et al. Deep multimodal pain recognition: A database and comparison of spatio-temporal visual modalities. In Proceedings of the 13th IEEE International Conference on Automatic Face and Gesture Recognition, FG 2018, Xi'an, China, 15–19 May 2018; pp. 250–257.
- 241. Celona, L.; Manoni, L. Neonatal facial pain assessment combining hand-crafted and deep features. *Lect. Notes Comput. Sci.* **2017**, *10590 LNCS*, 197–204.
- 242. Florea, C.; Florea, L.; Butnaru, R.; Bandrabur, A.; Vertan, C. Pain intensity estimation by a self-taught selection of histograms of topographical features. *Image Vis. Comput.* **2016**, *56*, 13–27.
- 243. Monwar, M.; Rezaei, S. Pain recognition using artificial neural network. In Proceedings of the Sixth IEEE International Symposium on Signal Processing and Information Technology (ISSPIT), Vancouver, BC, Canada, 29 January 2007; pp. 28–33.
- 244. Gholami, B.; Haddad, W.; Tannenbaum, A. Relevance vector machine learning for neonate pain intensity assessment using digital imaging. *IEEE Trans. Biomed. Eng.* **2010**, *57*, 1457–1466.
- 245. Gunes, H.; Piccardi, M. Automatic temporal segment detection and affect recognition from face and body display. *IEEE Trans. Syst. Man Cybern. Part B Cybern.* 2009, *39*, 64–84.
- 246. Joshi, J.; Goecke, R.; Parker, G.; Breakspear, M. Can body expressions contribute to automatic depression analysis? In Proceedings of the 2013 10th IEEE International Conference and Workshops on Automatic Face and Gesture Recognition, FG 2013, Shanghai, China, 22–26 April 2013.
- 247. Gioftsos, G.; Grieve, D. The use of artificial neural networks to identify patients with chronic low-back pain conditions from patterns of sit-to-stand manoeuvres. *Clin. Biomech.* **1996**, *11*, 275–280.
- 248. Dickey, J.; Pierrynowski, M.; Bednar, D.; Yang, S. Relationship between pain and vertebral motion in chronic low-back pain subjects. *Clin. Biomech.* **2002**, *17*, 345–352.
- 249. Mackey, S. Central neuroimaging of pain. J. Pain 2013, 14, 328–331.
- 250. Zunhammer, M.; Bingel, U.; Wager, T. Issues in Pain Prediction More Gain than Pain. *Trends Neurosci.* **2016**, *39*, 639–640.
- 251. Rosa, M.; Seymour, B. Decoding the matrix: Benefits and limitations of applying machine learning algorithms to pain neuroimaging. *Pain* **2014**, *155*, 864–867.
- Lee, J.; Mawla, I.; Kim, J.; Loggia, M.; Ortiz, A.; Jung, C.; Chan, S.T.; Gerber, J.; Schmithorst, V.; Edwards, R.; et al. Machine learning-based prediction of clinical pain using multimodal neuroimaging and autonomic metrics. *Pain* 2019, *160*, 550–560.
- Marquand, A.; Howard, M.; Brammer, M.; Chu, C.; Coen, S.; MourĂ£o-Miranda, J. Quantitative prediction of subjective pain intensity from whole-brain fMRI data using Gaussian processes. *NeuroImage* 2010, 49, 2178–2189.
- 254. Labus, J.; Van Horn, J.; Gupta, A.; Alaverdyan, M.; Torgerson, C.; Ashe-McNalley, C.; Irimia A.; Hong, J.Y.; Naliboff, B.; Tillisch, K.; et al. Multivariate morphological brain signatures predict patients with chronic abdominal pain from healthy control subjects. *Pain* **2015**, *156*, 1545–1554.
- 255. Boissoneault, J.; Sevel, L.; Letzen, J.; Robinson, M.; Staud, R. Biomarkers for musculoskeletal pain conditions: Use of brain imaging and machine learning. *Curr. Rheumatol. Rep.* **2017**, *19*, 5.
- 256. Ung, H.; Brown, J.; Johnson, K.; Younger, J.; Hush, J.; Mackey, S. Multivariate classification of structural MRI data detects chronic low back pain. *Cereb. Cortex* **2014**, *24*, 1037–1044.
- 257. Baliki, M.; Schnitzer, T.; Bauer, W.; Apkarian, A. Brain morphological signatures for chronic pain. *PLoS ONE* **2011**, *6*, e26010.
- 258. Robinson, M.; O'Shea, A.; Craggs, J.; Price, D.; Letzen, J.; Staud, R. Comparison of machine classification algorithms for fibromyalgia: Neuroimages versus self-report. *J. Pain* **2015**, *16*, 472–477.
- 259. Apkarian, A.; Sosa, Y.; Sonty, S.; Levy, R.; Harden, R.; Parrish, T.; Gitelman, D. Chronic back pain is associated with decreased prefrontal and thalamic gray matter density. *J. Neurosci.* **2004**, 24.

- 260. Schmidt-Wilcke, T.; Luerding, R.; Weigand, T.; et al.. Striatal grey matter increase in patients suffering from fibromyalgia—A voxel-based morphometry study. *Pain* **2007**, *132*.
- 261. Ferrari, M.; Mottola, L.; Quaresima, V. Principles, techniques, and limitations of near infrared spectroscopy. *Can. J. Appl. Physiol.* **2004**, *29*, 463–487.
- 262. Yücel, M.; Aasted, C.; Petkov, M.; Borsook, D.; Boas, D.; Becerra, L. Specificity of Hemodynamic Brain Responses to Painful Stimuli: A functional near-infrared spectroscopy study. *Sci. Rep.* **2015**, *5*, 9469.
- 263. Fernandez Rojas, R.; Huang, X.; Ou, K.L. A Machine Learning Approach for the Identification of a Biomarker of Human Pain using fNIRS. *Sci. Rep.* **2019**, *9*, 5645.
- 264. Eken, A.; Gäkacay, D.; Yilmaz, C.; Baskak, B.; Baltaci, A.; Kara, M. Association of Fine Motor Loss and Allodynia in Fibromyalgia: An fNIRS Study. *J. Mot. Behav.* **2018**, *50*, 664–676.
- 265. Han, I.; Won, S.; Kang, Y.; et al.. Cortical activation patterns of electrical pain stimulation using fNIRS. In Proceedings of the 2018 6th International Conference on Brain-Computer Interface (BCI 2018), GangWon, South Korea, 15–17 January 2018; pp. 1–3.
- 266. Podorin, A.; Shakhnov, V. MEMS sensors to assess pain sensitivity of human. In Proceedings of the 2nd International Telecommunication Conference Advanced Micro- and Nanoelectronic Systems and Technologies, Moscow, Russia, 1–2 June 2017; Volume 498.
- 267. Neves, E.; Pino, A.; Almeida, R.; Souza, M. Objective assessment of knee osteroarthritis in Parachuters by bioimpedance spectroscopy. In Proceedings of the 30th Annual International Conference of the IEEE Engineering in Medicine and Biology Society, EMBS'08— "Personalized Healthcare through Technology", Vancouver, BC, Canada, 20–25 August 2008; pp. 5620–5623.
- 268. Sathiyabama, G.; Krishnan, S.; Saravanan, D.; Kumaran, R. Bio-MEMS based sensor for acute and chronic pain management: An online measurement strategy for using biomedicai MEMS sensor in the assessment of pain intensity and study of inflammation. In Proceedings of the NCET NRES EM 2014: 2nd IEEE National Conference on Emerging Trends in New and Renewable Energy Sources and Energy Management, Chennai, India, 16–17 December 2014; pp. 227–230.
- 269. Braun, A.; Frank, S.; Wichert, R. The capacitive chair. Lect. Notes Comput. Sci. 2015, 9189, 397-407.
- Martínez-Martí, F.; Ocón-Hernández, O.; Martínez-García, M.; Torres-Ruiz, F.; Martínez-Olmos, A.; Carvajal, M.; Banqueri, J.; Palma, A. Plantar pressure changes and their relationships with low back pain during pregnancy using instrumented insoles. *J. Sens.* 2019, 2019, 1567584.
- 271. Luo, R.; Yih, C.C.; Su, K. Multisensor fusion and integration: Approaches, applications, and future research directions. *IEEE Sens. J.* 2002, *2*, 107–119.
- 272. Pombo, N.; Garcia, N.; Bousson, K. A data fusion model to evaluate computerized pain diaries on anxiety and depression assessment. In Proceedings of the 2016 IEEE International Black Sea Conference on Communications and Networking (BlackSeaCom 2016), Varna, Bulgaria, 6–9 June 2016.
- 273. Pombo, N.; Garcia, N.; Bousson, K.; Spinsante, S.; Chorbev, I. Pain assessment-can it be done with a computerised system? A systematic review and meta-analysis. *Int. J. Environ. Res. Public Health* 2016, 13, 415.
- 274. Gruss, S.; Geiger, M.; Werner, P.; Wilhelm, O.; Traue, H.; Al-Hamadi, A.; Walter, S. Multi-modal signals for analyzing pain responses to thermal and electrical stimuli. *J. Vis. Exp.* **2019**, 2019, e59057.
- 275. Johnson, A.; Yang, F.; Gollarahalli, S.; Banerjee, T.; Abrams, D.; Jonassaint, J.; Jonassaint, C.; Shah, N. Use of mobile health apps and wearable technology to assess changes and predict pain during treatment of acute pain in sickle cell disease: Feasibility study. *JMIR mHealth uHealth* **2019**, *7*, e13671.
- 276. Tracey, I.; Woolf, C.; Andrews, N. Composite Pain Biomarker Signatures for Objective Assessment and Effective Treatment. *Neuron* **2019**, *101*, 783–800.
- 277. Yang, F.; Banerjee, T.; Narine, K.; Shah, N. Improving pain management in patients with sickle cell disease from physiological measures using machine learning techniques. *Smart Health* **2018**, *7–8*, 48–59.
- 278. Yang, L.; Wang, S.; Jiang, X.; Cheng, S.; Kim, H.E. PATTERN: Pain Assessment for paTients who can't TEll using Restricted Boltzmann machiNe. *BMC Med. Inform. Dec. Mak.* **2016**, *16*, *73*.
- 279. Koskimäki, H.; Mönttinen, H.; Siirtola, P.; Huttunen, H.L.; Halonen, R.; R"oning, J. Early detection of migraine attacks based on wearable sensors—Experiences of data collection using Empatica E4. In Proceedings of the UbiComp/ISWC 2017—Adjunct Proceedings of the 2017 ACM International Joint Conference on Pervasive and Ubiquitous Computing and Proceedings of the 2017 ACM International Symposium on Wearable Computers, Maui, HI, USA, 11–15 September 2017; pp. 506–511.

- Haines, T.; Bowles, K.A. Cost-effectiveness of using a motion-sensor biofeedback treatment approach for the management of sub-acute or chronic low back pain: Economic evaluation alongside a randomised trial. *BMC Musculoskelet. Disord.* 2017, 18, 18.
- Biedermann, P.; Aleksejeva, J.; Mikkonen, J.; Wilde, D. Sensepack: An in-between wearable for body-backpack communication. In Proceedings of the ACM International Conference Proceeding Series, Oslo, Norway, 1–3 October 2018; pp. 922–927.
- 282. Gershon, R.; Cook, K. Use of Computer Adaptive Testing in the Development of Machine Learning Algorithms. *Pain Med.* **2011**, *12*, 1450–1452.
- 283. Singh, S.; Tiwari, S.; Abidi, A.; Singh, A. Prediction of pain intensity using multimedia data. *Multimed. Tools Appl.* **2017**, *76*, 19317–19342.
- 284. Singh, S.; Rastogi, V.; Singh, S. Pain Assessment Using Intelligent Computing Systems. *Proc. Natl. Acad. Sci. India Sect. A Phys. Sci.* 2016, *86*, 285–295.
- 285. Sikka, K. Facial expression analysis for estimating pain in clinical settings. In Proceedings of the 2014 International Conference on Multimodal Interaction, ICMI 2014, Istanbul, Turkey, 12–16 November 2014; pp. 349–353.
- 286. Susam, B.; Akcakaya, M.; Nezamfar, H.; Diaz, D.; Xu, X.; De Sa, V.; Craig, K.; Huang, J.; Goodwin, M. Automated Pain Assessment using Electrodermal Activity Data and Machine Learning. In Proceedings of the Annual International Conference of the IEEE Engineering in Medicine and Biology Society (EMBS), Honolulu, HI, USA, 18–21 July 2018; pp. 372–375.
- 287. Hung, M.; Bounsanga, J.; Liu, F.; Voss, M. Profiling Arthritis Pain with a Decision Tree. *Pain Pract.* 2018, *18*, 568–579.
- 288. Lötsch, J.; Sipilä, R.; Tasmuth, T.; Kringel, D.; Estlander, A.M.; Meretoja, T.; Kalso, E.; Ultsch, A. Machine-learning-derived classifier predicts absence of persistent pain after breast cancer surgery with high accuracy. *Breast Cancer Res. Treat.* 2018, 171, 399–411.
- 289. Lötsch, J.; Ultsch, A.; Kalso, E. Prediction of persistent post-surgery pain by preoperative cold pain sensitivity: Biomarker development with machine-learning-derived analysis. *Br. J. Anaesth.* **2017**, *119*, 821–829.
- 290. Fodeh, S.; Finch, D.; Bouayad, L.; Luther, S.; Ling, H.; Kerns, R.; Brandt, C. Classifying clinical notes with pain assessment using machine learning. *Med. Biol. Eng. Comput.* **2018**, *56*, 1285–1292.
- 291. Fodeh, S.; Finch, D.; Bouayad, L.; Luther, S.; Kerns, R.; Brandt, C. Classifying clinical notes with pain assessment. *Stud. Health Technol. Inform.* 2017, 245, 1261.
- 292. Gago-Veiga, A.; Pagán, J.; Henares, K.; Heredia, P.; González-García, N.; De Orbe, M.I.; Ayala, J.; Sobrado, M.; Vivancos, J. To what extent are patients with migraine able to predict attacks? *J. Pain Res.* **2018**, *11*, 2083–2094.
- 293. Soar, J.; Bargshady, G.; Zhou, X.; Whittaker, F. Deep learning model for detection of pain intensity from facial expression. *Lect. Notes Comput. Sci.* 2018, 10898 LNCS, 249–254.
- 294. Vaughn, M.; Cavill, S.; Taylor, S.; Foy, M.; Fogg, A. Interpretation and knowledge discovery from a MLP network that performs low back pain classification. *IEE Colloquium Knowl. Discov. Data Min.* **1998**, *2*, 1–4.
- 295. Akkaya, I.; Halici, U. Mouse face tracking using convolutional neural networks. *IET Comput. Vis.* **2018**, *12*, 153–161.
- 296. Lu, G.; Hao, Q.; Kong, K.; Yan, J.; Li, H.; Li, X. Deep convolutional neural networks with transfer learning for neonatal pain expression recognition. In Proceedings of the ICNC-FSKD 2018—14th International Conference on Natural Computation, Fuzzy Systems and Knowledge Discovery, Huangshan, China, 28–30 July 2018; pp. 251–256.
- Rodriguez, P.; Cucurull, G.; Gonalez, J.; Gonfaus, J.; Nasrollahi, K.; Moeslund, T.; Roca, F. Deep Pain: Exploiting Long Short-Term Memory Networks for Facial Expression Classification. *IEEE Trans. Cybern.* 2017, doi:10.1109/TCYB.2017.2662199.
- 298. Semwal, A.; Londhe, N. Automated Pain Severity Detection Using Convolutional Neural Network. In Proceedings of the International Conference on Computational Techniques, Electronics and Mechanical Systems (CTEMS 2018), Belgaum, India, 21–22 December 2018; pp. 66–70.
- Tavakolian, M.; Hadid, A. Deep binary representation of facial expressions: A novel framework for automatic pain intensity recognition. In Proceedings of the International Conference on Image Processing, ICIP, Athens, Greece, 7–10 October 2018; pp. 1952–1956.

- 300. Theagarajan, R.; Bhanu, B.; Angeles, D.; Pala, F. KnowPain: Automated system for detecting pain in neonates from videos. In Proceedings of the 2018 IEEE 18th International Conference on Bioinformatics and Bioengineering, BIBE 2018, Taichung, Taiwan, 29–31 October 2018; pp. 133–138.
- 301. Manini, T.; Mendoza, T.; Battula, M.; Davoudi, A.; Kheirkhahan, M.; Young, M.; Weber, E.; Fillingim, R.; Rashidi, P. Perception of older adults toward smartwatch technology for assessing pain and related patient-reported outcomes: Pilot study. *J. Med. Internet Res.* 2019, 21, e10044.
- 302. Chahal, D.; Kharb, L. Smart diagnosis of orthopaedic disorders using internet of things (IoT). *Int. J. Eng. Adv. Technol.* **2019**, *8*, 215–220.
- 303. Aung, M.; Bianchi-Berthouze, N.; Watson, P.; De C. Williams, A. Automatic recognition of fear-avoidance behavior in chronic pain physical rehabilitation. In Proceedings of the PERVASIVEHEALTH 2014: 8th International Conference on Pervasive Computing Technologies for Healthcare, Oldenburg, Germany, 20–23 May 2014; pp. 158–161.
- 304. The Emotion & Pain Project. EmoPain Dataset. Available online: http://www.emo-pain.ac.uk/ (accessed on 7 January 2020).
- 305. Automatic Recognition of Protective Behaviour in Chronic Pain Rehabilitation. Available online: https://discovery.ucl.ac.uk/id/eprint/1403044/4/Berthouze_UBI_Health_workshop_paper_press_AS1_ HA-nadia-Final.pdf (accessed on 7 January 2020).
- 306. Olugbade, T.; Aung, M.; Bianchi-Berthouze, N.; Marquardt, N.; Williams, A. Bi-Modal detection of painful reaching for chronic pain rehabilitation systems. In Proceedings of the 2014 International Conference on Multimodal Interaction (ICMI 2014), Istanbul, Turkey, 12–16 November 2014; pp. 455–458.
- 307. Ashraf, A.; Lucey, S.; Cohn, J.; Chen, T.; Ambadar, Z.; Prkachin, K.; Solomon, P. The painful face Pain expression recognition using active appearance models. *Image Vis. Comput.* **2009**, *27*, 1788–1796.
- 308. Lucey, P.; Cohn, J.; Prkachin, K.; Solomon, P.; Matthews, I. Painful data: The UNBC-McMaster shoulder pain expression archive database. In Proceedings of the 2011 IEEE International Conference on Automatic Face and Gesture Recognition and Workshops, FG 2011, Santa Barbara, CA, USA, 21–25 March 2011; pp. 57–64.
- Sikka, K.; Dhall, A.; Bartlett, M. Classification and weakly supervised pain localization using multiple segment representation. *Image Vis. Comput.* 2014, 32, 659–670.
- 310. Ringful LLC. Pain Care. Available online: https://appadvice.com/app/pain-care/347787779 (accessed on 7 January 2020).
- 311. Adibuzzaman, M.; Ostberg, C.; Ahamed, S.; Povinelli, R.; Sindhu, B.; Love, R.; Kawsar, F.; Ahsan, G. Assessment of Pain Using Facial Pictures Taken with a Smartphone. In Proceedings of the International Computer Software and Applications Conference, Taichung, Taiwan, 1–5 July 2015; Volume 2, pp. 726–731.
- 312. Crouthamel, M.; Quattrocchi, E.; Watts, S.; Wang, S.; Berry, P.; Garcia-Gancedo, L.; Hamy, V.; Williams, R. Using a ResearchKit smartphone app to collect rheumatoid arthritis symptoms from real-world participants: Feasibility study. J. Med. Internet Res. 2018, 20, e177.
- 313. Rabbi, M.; Aung, M.; Gay, G.; Reid, M.; Choudhury, T. Feasibility and acceptability of mobile phone-based auto-personalized physical activity recommendations for chronic pain self-management: Pilot study on adults. J. Med. Internet Res. 2018, 20, e10147.
- 314. Meawad, F.; Yang, S.Y.; Loy, F. Automatic detection of pain from spontaneous facial expressions. In Proceedings of the 19th ACM International Conference on Multimodal Interaction (ICMI 2017), Glasgow, UK, 13–17 November 2017; pp. 397–401.
- 315. Chauhan, U.; Reithinger, N.; MacKey, J. Real-time stress assessment through PPG sensor for VR biofeedback. In Proceedings of the 20th International Conference on Multimodal Interaction, ICMI 2018, Boulder, CO, USA, 16-20 Oct. 2018.
- 316. Smith-Nunes, G.; Shaw, A.; Neale, C. PainByte: Chronic pain and biomedical engineering through the lens of classical ballet & virtual reality. In Proceedings of the 12th International Conference on Tangible, Embedded, and Embodied Interaction, TEI 2018, Stockholm, Sweden, 18–21 March 2018; pp. 493–497.
- 317. Hogaboam, L.; Daim, T. Technology adoption potential of medical devices: The case of wearable sensor products for pervasive care in neurosurgery and orthopedics. *Health Policy Technol.* **2018**, *7*, 409–419.
- 318. Matheve, T.; Brumagne, S.; Demoulin, C.; Timmermans, A. Sensor-based postural feedback is more effective than conventional feedback to improve lumbopelvic movement control in patients with chronic low back pain: A randomised controlled trial. *J. NeuroEng. Rehabil.* **2018**, *15*, 85.

- 319. Dawes, T.; Eden-Green, B.; Rosten, C.; Giles, J.; Governo, R.; Marcelline, F.; Nduka, C. Objectively measuring pain using facial expression: Is the technology finally ready? *Pain Manag.* **2018**, *8*, 105–113.
- 320. Liu, P.; Yazgan, I.; Olsen, S.; Moser, A.; Ciftci, U.; Bajwa, S.; Tvetenstrand, C.; Gerhardstein, P.; Sadik, O.; Yin, L. Clinical valid pain database with biomarker and visual information for pain level analysis. In Proceedings of the 13th IEEE International Conference on Automatic Face and Gesture Recognition, FG 2018, Xi'an, China, 15–19 May 2018; pp. 525–529.
- 321. Dugailly, P.M.; Coucke, A.; Salem, W.; Feipel, V. Assessment of cervical stiffness in axial rotation among chronic neck pain patients: A trial in the framework of a non-manipulative osteopathic management. *Clin. Biomech.* **2018**, *53*, 65–71.
- 322. Subramaniam, S.; Doss, B.; Chanderasekar, L.; Madhavan, A.; Rosary, A. Scope of physiological and behavioural pain assessment techniques in children A review. *Healthc.Technol. Lett.* **2018**, *5*, 124–129.
- 323. Shanmugam, M.; Nehru, S.; Shanmugam, S. A wearable embedded device for chronic low back patients to track lumbar spine position. *Biomed. Res.* 2018, 2018, S118–S123.
- 324. Mamontov, D.; Polonskaia, I.; Skorokhod, A.; Semenkin, E.; Kessler, V.; Schwenker, F. Evolutionary algorithms for the design of neural network classifiers for the classification of pain intensity. *Lect. Notes Comput. Sci.* **2019**, *11377 LNAI*, 84–100.
- 325. Hu, B.; Kim, C.; Ning, X.; Xu, X. Using a deep learning network to recognise low back pain in static standing. *Ergonomics* **2018**, *61*, 1374–1381.
- 326. Amir, R.; Leiba, R.; Eisenberg, E. Anchoring the numeric pain scale changes pain intensity reports in patients with chronic but not with acute pain. *Pain Pract.* **2019**, *19*, 283–288.
- 327. Ostojic, K.; Paget, S.; Kyriagis, M.; Morrow, A. Acute and Chronic Pain in Children and Adolescents With Cerebral Palsy: Prevalence, Interference, and Management. *Arch. Phys. Med. Rehabil.* 2019, doi:10.1016/j.apmr.2019.08.475.
- 328. Kent, M.L.; Tighe, P.J.; Belfer, I.; Brennan, T.J.; Bruehl, S.; Brummett, C.M.; Buckenmaier, C.C.; Buvanendran, A.; Cohen, R.I. The ACTTION-APS-AAPM Pain Taxonomy (AAAPT) Multidimensional Approach to Classifying Acute Pain Conditions. J. Pain 2017, 18, 479–489.



© 2020 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).